IEEE INSTRUMENTATION & MEASUREMENT PUPPUPPUPPUS SOCIETY®

> Received 14 March 2022; revised 6 May 2022; accepted 8 May 2022. Date of publication 27 May 2022; date of current version 23 June 2022. The review of this paper was arranged by Associate Editor Subhas Chandra Mukhopadhyay.

> > Digital Object Identifier 10.1109/OJIM.2022.3178468

A Review on Sensing Technologies for High-Throughput Plant Phenotyping

ZHENYU MA¹⁰, RAKIBA RAYHANA¹⁰ (Student Member, IEEE), KE FENG¹⁰, ZHENG LIU¹⁰ (Senior Member, IEEE), GAOZHI XIAO²⁰ (Fellow, IEEE), YUEFENG RUAN³⁰, AND JATINDER S. SANGHA³⁰

¹School of Engineering, The University of British Columbia, Kelowna, BC V1V 1V7, Canada

²National Research Council (NRC) Canada, Ottawa, ON K1A 0R6, Canada

³Agriculture and Agri-Food Canada, Saskatchewan, SK S7N 0X2, Canada

CORRESPONDING AUTHOR: Z. LIU (e-mail: zheng.liu@uottawa.ca)

ABSTRACT The current epidemic, population growth, and decreasing arable lands lead to a severe food crisis, which calls for productive and efficient agricultural methods to ensure a sustainable food supply for mankind. Crop monitoring is considered to be a potential solution for the improvement of food production. Current crop monitoring combines agriculture methodologies with other advanced technologies, including sensing technology, geographical information systems (GIS), Internet of Things (IoT), information and communication technology (ICT), robotics, and drone techniques to increase production with low labor cost. The high-throughput plant phenotyping is crucial for crop monitoring on the data acquisition of large-scale crop characteristics. The high-throughput plant phenotyping studies aim to achieve fast and precise large-scale crop monitoring techniques with minimum environmental impact by applying special plant phenotyping platforms. The phenotyping platforms are integrated with various sensors and data communication systems, which can help to achieve automatic data acquisition and transmission. This paper reviews the current high-throughput plant phenotyping development in crop monitoring, including sensors, communication protocols, data management, and plant phenotyping platforms. State-of-art challenges are reviewed and discussed. Also, the paper provides discussions on the current situation, upcoming challenges, and possible future trends for researchers in this field.

INDEX TERMS High-throughput, plant phenotyping platforms, crop monitoring, sensing technology, communication protocols.

I. INTRODUCTION

A GRICULTURE has played a vital role in the development of the socio-economic condition and human civilization [1]. With the ever-increasing population, rapid urbanization, and the unexpected outburst of the pandemic, the supply chain of the food has been adversely affected [2], [3]. A recent report stated that by the year 2060, the world population would reach around 9.3 billion [4]. This crisis indicates that additional efforts and techniques are required to multiply the food production to feed the world's population [5]. Therefore, food production needs to be increased under the limited space condition to mitigate future food supply issues. One method is to employ crop monitoring practices with effective and efficient technologies to help overcome the food supply problems [6].

Crop monitoring is an effective agriculture method. It combines agriculture methodologies with advanced technologies, which include sensing technologies, geographical information systems (GIS), Internet of Things (IoT), information and communication technology (ICT), robotics, and drones techniques to increase the yield with relatively low labor cost [7]–[10].

The crop characteristics are influenced by various factors such as soil condition, temperature, crop moisture content, plant disease, etc. Even the same type of crop plants planted in a different field will have other characteristics. In traditional farming practice, the farmers need to visit the agriculture fields frequently to monitor crop conditions. However, this monitoring process is laborious and can consume up to 70% of farmer's time [11], [12]. Hence, advanced technologies such as sensor networks, ubiquitous computing [13], [14], and grid computing with satellite navigation services can improve the monitoring process and help make beneficial decisions for farmers.

Various sensors and their networks can constantly monitor the crop phenotyping with high accuracy, while the sensing data can be transferred to a secured platform through suitable communication protocols. Besides, robotic, unmanned ground vehicles (UGV) and unmanned aerial vehicles (UAV) are applied as a platform where sensors are installed to collect plant characteristics data. These data can be used to design decision-making platforms for the farmers [11]. To acquire and deal with large-scale plant characteristics data, studies regarding high-throughput plant phenotyping are essential [15]. Through high-throughput plant phenotyping, detailed and non-invasive large-scale plant information is enabled to be acquired throughout the plant life cycle for agricultural decision making and production increase [16]–[19].

This paper aims to aid the researchers in acquiring the knowledge of the sensing technologies for high-throughput plant phenotyping, the current challenges, and advancing the state-of-the-art. The paper summarizes the plant phenotyping parameters, communication protocols, and data management analyzes the current plant phenotyping sensors, and highlights the features of various sensor types, communication protocols, and data management methods. Furthermore, the paper reviews the up-to-date plant phenotyping platforms available for high-throughput plant phenotyping. This paper aims to provide an insight for the researchers to resolve the current challenges in high-throughput plant phenotyping. Compared with existing review papers, this study includes more comprehensive information on highthroughput plant phenotyping. It covers all the relevant details on high-throughput plant phenotyping, from basic plant phenotyping parameters and plant phenotyping sensors to sensing platforms, communication protocols, and data management.

The paper is organized as follows. Section II introduces the concept of plant phenotyping, the essential parameters, and communication protocols for plant phenotyping. Section III reviews the sensors used for high-throughput plant phenotyping. Section IV describes the up-to-date plant phenotyping platforms for plant phenotyping. Section V summarizes this paper and highlights the current challenges that need to be addressed and potential solutions.

II. OVERVIEW OF HIGH-THROUGHPUT PLANT PHENOTYPING

A. HIGH-THROUGHPUT PLANT PHENOTYPING

The term plant phenotyping refers to the plant appearance and performance, which can be written as $P = G \times E$,



FIGURE 1. The process of high-throughput plant phenotyping.

where P indicates the plant appearance and performance, Gindicates the plant genotype, E indicates the environmental variables. In other words, plant phenotyping is the result of genotype and environmental variables on the plant [20]. Plant phenotyping is related to the quantitative and qualitative plant traits, which is essential for studying plant response to environmental conditions. Hence, it does not only include the plant itself but also includes the response of plants in a specific environment, such as a specially controlled environment (greenhouse, etc.) or the natural environment [21]. The target of plant phenotyping is to provide an analysis of plant traits to acquire its influence factors, which can help the plants to have a better yield in limited growing environments [20]. With the rapid development of electrical, computer science, and sensor technologies, high-throughput plant phenotyping has been a trendy field of study [22]. High-throughput plant phenotyping means automatic, noninvasive phenotyping systems, which have the ability for automated data acquisition, processing, analysis, and visualization [21]. Currently, the requirements for high-throughput plant phenotyping research are increasing because it plays a crucial role in crop monitoring [20], [23]. Fig. 1 illustrates the process of high-throughput plant phenotyping.

B. SENSING PARAMETERS FOR HIGH-THROUGHPUT PLANT PHENOTYPING

The plant phenotyping parameter is an essential part for highthroughput phenotyping studies. Recent studies show that the phenotyping parameters of economic crops (including barley [24], wheat [25], sorghum [26], maize [27], tomato [28], bean [29] and cotton [30]) are taken into account. These plant phenotyping parameters can be grouped into two categories: morphometric and physiological [22]. Fig. 2 presents some

IEEE INSTRUMENTATION & MEASUREMENT INTERPOLITION SOCIETY®



of important plant phenotyping parameters in morphometric and physiological phenotyping.

Morphometric phenotyping is a mathematical description of biological forms based on the geometric definition of an object's size and shape [31]. Detailed morphometric information can help understand the effect of plant genes and the environmental condition on plant development. In morphometric phenotyping parameters, the plant height, stalk length, stem diameter, and leaf morphology (leaf length/width/area/angle) are crucial parameters because they are directly related to the plant growth [22]. The limitation of morphometric phenotyping parameters is that these parameters only stay on the external plant phenotyping, which is not enough to fully represent plant conditions like water or nutrient content [21].

Physiological phenotyping is about all the internal activities of plants, such as photosynthesis, plant nutrition, environmental stress, etc. Although many high-throughput plant phenotyping kinds of research still focus on morphometric phenotyping, the measurement of physiological phenotyping has been expected as an essential part of studying plant gene and plant performance relationships. In a physiological view, many abiotic (water, light, radiation, temperature, humidity, and soil condition) and biotic (human influence, pathogens, and disease) factors will affect plant cellular responses, ultimately determining plant performance. The physiological phenotyping parameters contain high dimensional data such as biomass, NDVI, chlorophyll, plant/water stress, leaf water content, disease resistance, and plant nutritional status. The limitation of physiological phenotyping is that the detection processes are relatively more sophisticated and expensive than morphometric phenotyping [21], [32].

C. COMMUNICATION PROTOCOLS FOR HIGH-THROUGHPUT PLANT PHENOTYPING

Data communication is another essential part of highthroughput plant phenotyping because it enables sensing data transmission. Through communication protocols, sensors can exchange the plant phenotyping data over the network. Different protocols have various frequency bands, data rates, transmission ranges, energy consumption, and costs [33]. Therefore, the choice of communication protocols



depends on actual applications. The commonly used communication protocols include IEEE 802.11 (WiFi), WiMax, BlueTooth, Cellular, LoRaWAN, and NB-IoT. The following contents briefly describe these communication protocols. Table 1 presents the comparison of these communication protocols.

and in-depth trait analysis)

1) IEEE 802.11 (WiFi) is a series of the Wireless Local Area Network (WLAN) communication standards, including the IEEE 802.11, 802.11a, 802.11b, 802.11g, 802.11n, 802.11ac, 802.11ax standards. The state of art standard is 802.11ax (also called WiFi 6), which was developed in May 2014 to improve the throughput-per-area in high-density scenarios [34]. Different standards have various technical parameters; the frequency band of WiFi varies from 5 to 60 GHz, while the rated speed varies from 0.25 Mb/s to 6.75 Gb/s. The latest standard reaches a significant rate speed of 6.75 Gb/s, with 2.4GHz and 5GHz frequencies. Besides, the current range of WiFi covers from 20 m to 100 m [33], [34]. With relatively stable and high data rates, WiFi has become a universal solution for many applications, including crop monitoring [35]. The study [36] applied WiFi to do the data transmission in their system, which is an imaging technology and cloud technology-based system. Their system can achieve real-time crop detection and crop growth status recording. The result of this work shows that this system can complete real-time detection and data transmission in the field of crop monitoring [36]. Zhou et al. [37] introduced a large-scale automatic field phenotyping platform, CropQuant, which is a low-cost Internet of Things (IoT) powered system. By using high-resolution time-lapse photography, in-field evaluation, and a WiFi-based data exchange system, this platform can achieve continuous crop monitoring [37]. Fig. 3 shows the principle of the CropQuant system.

Frequency Band	Data Rate	Transmission Range	Energy Consumption	Cost	References
			Consumption		
5 GHz – 60 GHz	0.25 Mb/s - 6.75 Gb/s	20 m – 100 m	High	High	[33, 34]
2 GHz – 66 GHz	Mobile: 100 Mb/s,	50 km	Medium	High	[33]
	Fixed: 1 Gb/s				
2.4 GHz	1 Mb/s – 24 Mb/s	8 – 10 m	Very Low	Low	[33, 41]
865 MHz, 2.4	3G: 200 kbps,	Whole Cellular Area	Medium	Medium	[33, 39, 43,
GHz,	4G: 5 – 16 Mbps,				44]
5G (5.8 GHz, 60	5G: 100 Mbps – 1				
GHz)	Gbps				
868/915 MHz,	4 kbps	30 km	Very Low	High	[39, 49]
2.4 GHz					
800/900/1800	29 kbps	10 km	Low	High	[39, 46,
MHz (EU)					50]
	Frequency Band 5 GHz - 60 GHz 2 GHz - 66 GHz 2.4 GHz 865 MHz, 2.4 GHz, 5G (5.8 GHz, 60 GHz) 868/915 MHz, 2.4 GHz 868/915 MHz, 2.4 GHz 800/900/1800 MHz (EU)	Frequency Band Data Rate 5 GHz - 60 GHz 0.25 Mb/s - 6.75 Gb/s 2 GHz - 66 GHz Mobile: 100 Mb/s, Fixed: 1 Gb/s 2.4 GHz 1 Mb/s - 24 Mb/s 865 MHz, 2.4 3G: 200 kbps, 4G: 5 - 16 Mbps, 5G (5.8 GHz, 60 GHz) 5G 868/915 MHz, 2.4 GHz 4 kbps 800/900/1800 MHz (EU) 29 kbps	$\begin{array}{ c c c c c c } \hline Frequency Band & Data Rate & Transmission Range \\ \hline 5 \ GHz - 60 \ GHz & 0.25 \ Mb/s - 6.75 \ Gb/s & 20 \ m - 100 \ m \\ \hline 2 \ GHz - 66 \ GHz & Mobile: 100 \ Mb/s, \\ Fixed: 1 \ Gb/s & 50 \ km \\ \hline 2.4 \ GHz & 1 \ Mb/s - 24 \ Mb/s & 8 - 10 \ m \\ \hline 865 \ MHz, 2.4 & 3G: 200 \ kbps, \\ GHz, & 4G: 5 - 16 \ Mbps, \\ 5G \ (5.8 \ GHz, 60) & 5G: 100 \ Mbps - 1 \\ GHz & Gbps & \hline \\ 868/915 \ MHz, & 4 \ kbps & 30 \ km \\ \hline 2.4 \ GHz & 10 \ Mb/s & 29 \ kbps & 10 \ km \\ \hline \hline 800/900/1800 & 29 \ kbps & 10 \ km \\ \hline \end{array}$	Frequency BandData RateTransmission RangeEnergy Consumption $5 \text{ GHz} - 60 \text{ GHz}$ $0.25 \text{ Mb/s} - 6.75 \text{ Gb/s}$ $20 \text{ m} - 100 \text{ m}$ High $2 \text{ GHz} - 66 \text{ GHz}$ Mobile: 100 Mb/s, Fixed: 1 Gb/s 50 km Medium 2.4 GHz 1 Mb/s - 24 Mb/s $8 - 10 \text{ m}$ Very Low 865 MHz , 2.4 $3G: 200 \text{ kbps}$, $GHz,GHz,3G: 200 \text{ kbps},4G: 5 - 16 Mbps,5G: 100 \text{ Mbps} - 1GHz)Whole Cellular AreaBesMedium868/915 \text{ MHz},2.4 \text{ GHz}4 \text{ kbps}30 \text{ km}Very Low800/900/1800MHz (EU)29 \text{ kbps}10 \text{ km}Low$	Frequency BandData RateTransmission RangeEnergy ConsumptionCost $5 \text{ GHz} - 60 \text{ GHz}$ $0.25 \text{ Mb/s} - 6.75 \text{ Gb/s}$ $20 \text{ m} - 100 \text{ m}$ HighHigh $2 \text{ GHz} - 66 \text{ GHz}$ Mobile: 100 Mb/s, Fixed: 1 Gb/s 50 km MediumHigh 2.4 GHz 1 Mb/s - 24 Mb/s $8 - 10 \text{ m}$ Very LowLow 865 MHz , 2.4 $3G: 200 \text{ kbps}$, $4G: 5 - 16 \text{ Mbps}$, $5G (5.8 \text{ GHz}, 60)$ $3G: 100 \text{ Mbps} - 1$ GHz MediumMedium $868/915 \text{ MHz}$, 2.4 GHz 4 kbps 30 km Very LowHigh $800/900/1800$ MHz (EU) 29 kbps 10 km LowHigh

TABLE 1. Comparison of communication protocols.

- 2) WiMAX (Worldwide Interoperability for Microwave Access) is a high-speed wireless data communication protocol based on IEEE 802.16 standard [38]. It provides a data rate from 1.5 Mb/s to 1 Gb/s. The new standard version 802.16m can reach the data rate of 100 Mb/s through mobile stations and up to 1 Gb/s through fixed stations [33]. The WiMAX frequency band varies from 2 to 66 GHz, while the covering range can reach a maximum of 50 km. However, these features are also accompanied by relatively high cost and medium energy consumption [39]. The study [40] applied Wimax-based technology to develop an agriculture monitoring system, which can integrate the wireless sensor network and send the data to the central server [40].
- 3) Bluetooth is a kind of low energy consumption and low-cost wireless data communication protocol based on the IEEE 802.15.1 standard. The Bluetooth standard can build a communication link between portable devices, such as laptops, over 8 to 10 m. It provides a personal area network (PAN) communication on the 2.4 GHz band. The data rate ranges from 1 Mb/s to 24 Mb/s in different standard versions [33], [41]. Kuhlgert et al. [42] designed a device called MultispeQ to collect large-scale, high-quality field data. The MultispeO can provide useful plant phenotyping such as chlorophyll and pigment information. These plant phenotyping data can be transmitted from the MultispeQ to mobile phones or laptops and be saved to their data management platform. The communication method between the MultispeQ and other devices is Bluetooth or micro-USB. Their study shows that this system can measure, send, store and analyze the plant phenotyping data [42].
- 4) Cellular, also called mobile communication, is suitable for portable devices which have the requirement of high data transmission rate. There are four generations of cellular standards, which include second generation (2G including GSM and CDMA), third generation (3G including UMTS ad CDMA2000), fourth generation (4G including LTE), fifth-generation (5G

NR) [33], [43]. Based on this kind of communication protocol, portable devices can communicate over cellular networks. Data rate of cellular range from 9.6 Kb/s (2G) to 1 Gbps (5G) [33], [44]. The study [45] developed a multi-billion pixel ("gigapixel") image sensor system, which can detect the high-throughput phenotyping in field settings and transmit the images to a remote server by cellular connection. The authors state that the data rate of 4G is enough for gigapixel images uploading in their system [45].

- 5) LoRaWAN is a new communication protocol for longdistance data communication. It has the advantage of long-range data transmission. The LoRaWAN network applies a new technique to achieve multiple messages received through various channels during data transmission. By using the new design, LoRaWAN can increase the data communication range and network size while keeping a long battery life [46]. To sum up, LoRaWAN has a high capacity and low energy consumption with limited data rates. Its data rates can reach the requirements of standard agriculture sensors [39]. Singh et al. [47] developed a LoRaWANbased sensor network to monitor tomato crops in the greenhouse. The LoRaWAN will transmit the raw sensor data to the LoRaWAN gateway and forward it to a working network. Then a data parser will subscribe to the data from the working network and use the data for visualization and store [47].
- 6) NB-IoT is the Low Power Wide Area Network (LPWAN) radio technology standardized by 3GPP. It focuses on indoor, low energy consumption, low expense, and high connection density communication work. Although it applies the subset of LTE (Long Terminal Evolution) standard, it removes some features of LTE to reduce the energy consumption and cost [39], [46]. The study [48] stated an intelligent plant temperature control system based on NB-IoT low-rate narrow bandwidth network. The system can do the data transmission within an indoor distance and achieve automatic regulation of plant growth temperature. The result of this study shows that the

9500121





(a) CCD sensor

(b) CMOS sensor

FIGURE 4. CCD and CMOS sensors [73].

system can stably regulate a beneficial condition for plant growth [48].

D. DATA MANAGEMENT FOR HIGH-THROUGHPUT PLANT PHENOTYPING

The massive data generated from the high-throughput plant phenotyping process must be managed. The managed data will be used for processing and analysis to obtain the plant phenotyping results. The database management system (DBMS) needs to be applied to maintain and control access to plant phenotyping data [51]. The database management systems can be divided into relational and non-relational. A relational database is a set of tables that store and provide data access. The row in the table is a unique ID, and corresponding columns contain attributes of the data. The non-relational database does not apply tables as a data storage structure. It provides a solution for largescale data storage requirements of the Internet. According to the difference in structure and application scene, the nonrelational database can be divided into key-value stores, document stores, and XML databases [52]. The frequently-used relational database management systems include MySQL, PostgreSQL, Microsoft SQL Server, Oracle, and Microsoft Access. The frequently-used non-relational database management systems include MongoDB, Redis, and CouchDB. Most real-time plant phenotyping data can use relational databases since they are suitable for the representation of tables [53]. Schmidt et al. [54] developed a distributed information system called Phenomics to manage massive plant phenotyping data. In the developed system, a relational database management system PostgreSQL was applied to save plants' treatments and measurements data. The plant data are divided into various parts in database tables (like experiments, plants, locations) [54]. Reynolds et al. [55] developed a distributed plant phenotyping management system called CropSight. In the developed system, a relational database management system, MySQL, was applied for storing various data, including images, weather data, and experimental settings [55].

With the enormous development of cloud storage, various organizations have started to upload their data to cloud storage since it can provide a low-cost and convenient method for large-scale data storage. Through the Internet, users can access the stored crop data anytime and anywhere [56]. The current commercial cloud storage systems include Amazon S3, Microsoft Azure, Alibaba Cloud, etc. [56].

III. SENSORS FOR HIGH-THROUGHPUT PLANT PHENOTYPING

Currently, many sensors have become essential elements for plant phenotyping parameters measurement as it could provide fast and noninvasive information of plants [20]. Many advanced technologies have been integrated with sensors to perform the plant phenotyping work, such as image techniques [57], machine learning [58] and communication techniques [59]. Studies show that the traditional and even new sensors are integrated for plant phenotyping parameters measurement, which includes visiable spectrum image sensor [60], thermal image sensor [61], NIR image sensor [24], multi-spectral image sensor [62], hyperspectral image sensor [63], Light detection and ranging (LiDAR) [64], ultrasonic sensor [65], fluorescence sensor [66], depth image sensor [67] and etc. The following subsections discusses different types of sensors that are being widely used for plant phenotyping.

A. VISIBLE SPECTRUM IMAGE SENSOR

The visible spectrum is the segment of the electromagnetic spectrum that human eyes can view-typically, visible spectrum wavelengths about 400 to about 780 nm [68]. A visible spectrum image sensor can collect visible light and create images based on red, green, and blue wavelengths (RGB) for accurate color representation [69]. The visible spectrum image sensor mostly consists of charge-coupled device (CCD) silicon sensors and active-pixel sensors (CMOS) [69]. Photons fall on the sensor surface and generate a charge, which can be transferred into a visible copy on the device. Therefore, a visible spectrum image sensor can obtain visible radiation of an object to form an RGB image, which can be used to present a plant's physical information (morphological and texture features), biotic stress, and plant status [22]. Afterimage processing and analysis, various plant phenotyping parameters (including plant height, stalk height, stem diameter, leaf length/width/area/angle and biomass) could be acquired [60]. Visible spectrum image sensors are low-cost, simple, and ubiquitous. Moreover, the technical parameters of visible spectrum image sensors such as light sensitivity, image resolution, and ability to focus have been significantly improved every year [70]. Besides, with the rapid development of computer vision, implementing computer vision technology to extract helpful information from RGB images has become a key method for plant phenotyping studies. It can automate the data acquisition of the plant phenotyping parameters. This can help save time and cost for the farmers because they do not have to go to the fields for plant monitoring and measurement. In addition, the applications of deep learning and convolutional neural networks in plant image analysis have greatly improved the high-throughput plant phenotyping work efficiency [71]. Furthermore, highthroughput platforms have been greatly developed and can provide a controlled environment to study the plant responses to the environment [72]. Therefore, visible spectrum image sensors in high-throughput plant phenotyping studies are gaining popularity. Fig. 4 presents the CCD and CMOS



FIGURE 5. Visible spectrum image sensor principle.

sensor. Fig. 5 shows the visible spectrum image sensor principle. When the visible spectrum image sensor starts to work, then the light of the plant will enter through the lens. Then, photons fall on the CCD/CMOS sensor surface and generate a charge. That charge will be converted to a voltage, amplified, and processed by an analog to digital Converter (AD converter). After that, the image processing unit will process the digital signal and generate a digital image.

One methodology for applying visible spectrum image sensors in high-throughput platforms is using multiple image sensors to take many images of plants from different views. This can help to acquire consistent and accurate phenotyping parameters. The research [74] applied the visible spectrum image sensors based on LemnaTec 3D Scanalyzer (LemnaTec, GmbH, Wuerselen, Germany) high-throughput platform to take RGB images of barley plants in three different views (one top view image and two side-view images with a 90-degree horizontal rotation). These RGB images were used to predict barley growth and estimate the biomass. The result of this research shows that there is a high correlation between visible spectrum image-based biomass estimation and actual plant biomass. This indicates that the visible spectrum image sensor-based phenotyping system had good performance on high-throughput plant phenotyping estimation [74]. Similarly, [24] developed an integrated framework for high-throughput plant phenotyping data analysis. This new framework can extract plant phenotyping parameters from noninvasive images regularly. The multiple image sensors were also used to take three different views of RGB images to measure the status and biomass of the plant for the barley plants and store the data. This study shows that the framework data are precious to the plant development and biomass variation. It also helps to quantify further plant growth and performance features [24]. The study [66] also used the LemnaTec 3D Scanalyzer system (LemnaTec GmbH, Aachen, Germany) to take three different views of RGB images of two different salinity tolerance rice cultivars for plant assessment. The RGB images were used for plant height; the images were also applied to estimate approximation biomass. The result from this study shows that the projected shoot area obtained by RGB image analysis has a strong positive correlation with biomass on older plants [66]. Similar study methods are also applied to the sorghum [75] and wheat plants [76].

Combining the digital image sensor and UAV is another popular methodology in plant phenotyping studies. The study [77] built a high-throughput plant phenotyping platform based on UAV for soybean yield estimation and crop maturity prediction. The designed system applied a fast and inexpensive method for plant evaluation and improved crop breeding efficiency. The result of this study shows that the traditional yield estimation method could be greatly improved, and high accuracy (93%) has been achieved in soybean maturity classification [77].

The studies above did not mention the relationship between plant phenotypes and genomics. Therefore, the study [78] developed a high-throughput rice phenotyping facility (HRPF) and collaborated it with genome-wide association studies (GWAS). A visible spectrum image sensor-based high-throughput platform can obtain morphology parameters (including plant height, green leaf area, plant compactness) and rice biomass. Plant compactness is a new trait that traditional plant phenotyping methods cannot be defined and obtain among these morphology parameters. Their novel work shows that the new high-throughput plant phenotyping method performs better than the conventional plant phenotyping method. It can provide valuable plant gene identification information. Besides, the use of multiple phenotyping tools includes in-depth information and insight into the relationship between plant phenotyping and genetic architecture [78].

One big challenge of the visible spectrum image sensorbased plant phenotyping is the difficulty of object distinction between plant and background with similar plant colors such as grapevine and field. A study conducted by [79] used a normally visible spectrum image sensor to compute grapevine leaf surface areas and fruit-to-leaf ratios. They segmented RGB images into four classes (leaf, stem, grape, and background) by depth reconstruction, color classifier, and edge detector. The segmented images help to acquire the leaf surface areas and fruit-to-leaf ratio. The study result shows a good determination coefficient and minor error compared with the traditional method. This research found a possible way to solve the challenge of object distinction in plant phenotyping. This advance may improve the development of high-throughput plant platforms and image acquisition [79].

Although significant progress has been made so far for applying visible spectrum image sensors on plant



FIGURE 6. Infrared spectrum principle.

phenotyping, such as the visible spectrum image sensorbased high-throughput platform and image analysis, the limitation of digital image sensors still needs to be considered. Firstly, the visible spectrum image sensor is limited to visual spectral bands [80], which means it may lose some spectrum information such as infrared spectrum. Various spectra can provide different details about the plant phenotyping, such as plant water content and plant temperature [72]. This will lead the high-throughput platforms to apply more sensors for efficient plant information. Secondly, the performance of the visible spectrum image sensor is easily affected by the varying light conditions [22]. The study [81] explored a method to use the visible spectrum image sensor scene mode to measure nitrogen nutrition of corn without additional artificial illumination. Therefore, the development of visual spectrum image sensor scene mode may be a potential solution to the problem of light effect.

B. INFRARED SPECTRUM IMAGE SENSOR

The wavelength of the infrared spectrum is about 780 nm - 1000 μ m. The light wavelength is just outside the visible range of the human eyes, and it can offer unique details about plants [82]. Based on the wavelength, the infrared spectrum can be divided into five parts: near-infrared (NIR), shortwave infrared (SWIR), mid-wave infrared (MWIR), long-wave infrared (TIR/LWIR), and far-infrared (FIR). Fig. 6 shows the infrared spectrum principle. Some properties of the plant are related to infrared light, which could be potentially helpful for plant phenotyping analysis [72].

Near-infrared (NIR) image sensor contains Indium Gallium Arsenide (InGaAs) sensors. Because of its spectral sensitivity, the NIR image sensor can detect near-infrared and short-wave infrared spectrum, which is useful for measuring leaf water content [72]. Compared with visible light, NIR and SWIR lights could be greatly reflected by plant tissue, and the reflection is influenced by leaf thickness, which is related to leaf water content. Many studies have applied the NIR image sensor in leaf water content measurement and plant drought response. Fig. 7 presents the reflectance under different leaf water content. When the leaves have a regular water content (healthy), the reflection of near-infrared light is larger than blue, green, and red light. When the water content of leaves decreases (stressed), the reflection of nearinfrared light will gradually reduce. After the leafs withered (dead), the near-infrared reflection will be almost the same



FIGURE 7. An illustration of leaf reflectance [83].



FIGURE 8. NIR image sensor [84].

as blue, green, and red light. Fig. 8 shows the image of the NIR sensor.

The research [85] used a NIR image sensor to detect plant water content. Their study is based on a standard plant feature that leaf dehydration will decrease NIR light's absorption. The result of the study shows that some wavelengths (970/1200 nm) have lower sensitivity due to small absorption troughs, while some wavelengths (1450/1930/2500 nm) have much higher sensitivities of leaf water indices to leaf water content [85]. The research [24] applied multiple image sensors to extract phenotyping parameters of barley by noninvasive imaging constantly. In this system, the NIR image sensor was used to obtain the plant water content [24]. The study [86] introduced that the morphometric and plant drought response of grapevines can be acquired through RGB and NIR image analysis. The result of this study shows that NIR is related to the levels of plant drought and enhances





(a) Multi-spectral sensor (b) Multi-spectral camera

FIGURE 10. Multi-spectral sensor [93], [94].

FIGURE 9. Thermal infrared image sensor [89].

the role of the NIR image sensor in plant drought response study [86].

NIR image sensor also has similar limitations as digital image sensor. It is limited to specific bands, and the other wavelength information will be lost in the NIR image. The potential solution is to apply multiple image sensors together or use the hyperspectral image sensors instead of NIR image sensors [72].

Thermal infrared (TIR) image sensor can detect the longwave infrared (LWIR) spectrum. Thermal infrared imaging is a remote, non-invasive technology that can provide temperature mapping of an object. Thermal infrared imaging has been widely applied in many areas such as aerospace, automotive industry, medicine, and security. All the objects above 0 °C will emit infrared rays, which can be represented as surface temperature. The higher temperature of the surface will have a greater infrared radiation intensity.

Therefore, thermal infrared imaging technology can convert the radiation into temperature data. The main advantages of thermal imaging are it can be operated remotely. It is noninvasive, easy to handle, and has higher accuracy than other sensors. Besides, thermal infrared imaging can acquire a temperature map of the captured region with a fast response, not the simple spot temperature data. In addition, unlike other imaging systems, thermal infrared imaging does not have a strict requirement for illumination, which means it can effectively work in a dark environment. The thermal infrared image sensor is comprised of infrared detectors' signal processing parts. The infrared detectors can detect the intensity of radiation and convert it into an electrical signal, which will be translated into a thermal image in the signal processing part [87]. Thermal infrared images can be used to represent plant temperature, plant water stress, and plant quality [72], [88]. Fig. 9 presents the thermal infrared sensor and a thermal infrared image.

Matos *et al.* [90] used thermal infrared image sensors to study the influence of environment temperature and light changes on the plants. They used different temperature and light environment conditions to test the plant's growth rate. This study shows that temperature might primarily influence plant growth rate. In addition, the temperature variation did not affect the size of the leaf cell length [90]. Crusiol *et al.* [91] applied the thermal infrared image to evaluate the temperature of soybean plants and their water status. A thermal infrared image sensor was integrated with an unmanned aerial vehicle (UAV) to capture the thermal infrared image of plants. This work shows the relationship between plant thermal infrared behavior and soil water availability. The lower soil moisture conditions will lead to higher plant temperatures. Besides, this work shows the value of applying UAV and thermal infrared image sensors for plant water monitoring and crop management [91].

The thermal infrared image sensor plays a vital role in the temperature mapping of the plant. Due to its powerful advantages, various thermal infrared image sensors have already been applied in plant phenotyping. However, the thermal infrared image sensor is easily affected by environmental conditions, and it is hard to detect the difference in minor temperature variations [80]. The study [87] introduces that the thermal infrared image sensor can be divided into the uncooled and cooling device. The uncooled device can be operated at room temperature, and most studies used it. Although these uncooled devices are inexpensive, their resolution and image quality is lower than the cooling device. The cooled thermal infrared imaging devices have high resolution and can detect a difference of slight temperature variations (0.1 °C). However, the cooled thermal infrared imaging devices are expensive and need to be maintained under 0 °C. With the development of new materials, the cooled thermal infrared imaging devices may solve the current problems of the thermal infrared image sensor and become the typical sensor for plant phenotyping measurement [87].

C. MULTI-SPECTRAL IMAGE SENSOR

Multi-spectral imaging is primarily developed for spacebased imaging. It has been successfully applied in many fields, such as environment, national defense, and agriculture. The multi-spectral image sensor can detect light with a few discrete wide spectral bands (generally 3 to 10). Therefore, it can provide more detailed spectral information. In the agricultural field, multi-spectral image sensors are usually used to detect the information of visible spectrum and near-infrared spectrum, which can be used to measure some specific plant phenotyping such as plant stress, water stress, disease detection and normalized difference vegetation index (NDVI) [80], [92]. Fig. 10 presents the images of the multi-spectral sensor and camera. Fig. 11 presents the multi-spectral principle.



FIGURE 11. Multi-spectral and hyperspectral principle.

Andrade-Sanchez et al. [95] used a multi-spectral crop canopy image sensor to measure the cotton canopy reflectance. The multi-spectral image sensor has three channels for detecting the visible and NIR spectrum. The canopy reflectance data in the infrared spectrum regions were implemented to calculate the normalized difference vegetation index (NDVI) [95]. Zaman-Allah et al. [96] used an unmanned aerial platform (UAP) equipped with sensors to get multi-spectral images of maize's performance under low nitrogen stress. The study results show that the study indicates that UAP based platform can effectively assess crop performance under low nitrogen stress. The NDVI data from multi-spectral images have a strong correlation with ground-measured data [96]. Patrick et al. [97] applied multispectral image sensor for tomato spot wilt disease resistance in peanuts research. Their study used a quadcopter with a multi-spectral image sensor to get multi-spectral images for high-throughput plant phenotyping of tomato spot wilt disease resistance among twenty different genotypes of peanuts. The multi-spectral images were analyzed and processed into several vegetation indices (each plot has a further distribution of pixel intensities), which can be applied to develop a model for plant disease assessment by working with manually acquired data. The result of the study shows it can save time and provide more reliable results than human visual work [97]. Kumar et al. [98] used a multi-spectral image sensor mounted on an unmanned aerial vehicle (UAV) to get the near-infrared, green, and red band images, which were applied to achieve the vegetative index for maize crop health monitoring. The study results show that this method successfully detects the water-stressed area. The irrigation process and crop health monitoring have been optimized through this study [98]. Mardanisamani et al. [99] applied a multi-spectral image sensor with five spectral channels (red, blue, green, near the infrared, and red edge) on a quadcopter to get the images of wheat and canola from two breeding field trials. A deep convolutional neural network was used on these images to build a detection model for plant lodging. The result of this study shows that the model can achieve

real-time plant lodging classification and reduce the work cost [99].

ICCC

IEEE INSTRUMENTATION & MEASUREMENT

The limitation of the multi-spectral image sensor is that it is limited to a few spectral bands. Few spectral bands may not be enough for some applications. A hyperspectral image sensor that provides hyperspectral details might solve this problem because the hyperspectral images have more measured wavelength bands than multi-spectral images [80]. Besides, employing a combination of image sensors could also be a potential solution for this limitation.

D. HYPERSPECTRAL IMAGE SENSOR

Hyperspectral images have hundreds of narrow spectral bands, which means they can provide enough spectral information for object analysis. The main difference between the multi-spectral and the hyperspectral image sensor is the number of wavebands and bandwidth being detected. Hyperspectral image sensor provides a potential method for more accurate and detailed plant phenotyping information than any other type of sensor. With the development of hyperspectral image sensors, it has become one of the most powerful sensors in the field of remote sensing [100]. Therefore, the hyperspectral image sensor is turning into a gradually popular method for high-throughput plant phenotyping as it can provide valuable insights into the plant conditions such as water content, plant stress, and chlorophyll [101]. In high-throughput plant phenotyping studies, the hyperspectral image sensor is usually applied to plant stress, plant quantitative assessment, and plant condition monitoring [102]. Fig. 11 presents the hyperspectral principle. Fig. 12 presents the images of the hyperspectral image sensor and camera.

Seiffert et al. [105] applied the hyperspectral image sensors to do the quantitative assessment of various genetically different tobacco varieties under different growing environments. The artificial neural networks were used to analyze the hyperspectral images in their study. This study shows that the different genotypes and growing conditions will lead to the difference in plant spectrum [105].



(a) Hyperspectral image sensor (b) Hyperspectral camera



FIGURE 13. Depth Image Sensor [114].

FIGURE 12. Hyperspectral sensor [103], [104].

As the plant condition monitoring area, [106] stated a method to classify the nutrition state of the crop through hyperspectral imaging. In their research, artificial neural networks were used to predict a plant's nutrition condition based on leaf age or intra-leaf pixel position. The result shows that their method can boost classification performance, where leaf age had a more substantial impact. Besides, the result also indicates that different spectral bands have different weights on plant nutrition condition prediction [106].

Römer et al. [107] applied unsupervised learning on the hyperspectral images, which are acquired by a hyperspectral image sensor SOC-700. Their work model can compare the plant spectrum with observed typical spectra to measure if the plant is suffering from stress. The result of this study tests the plant drought stress detection in cereals and provides a method to visualize the plant stress [107]. Mahlein et al. [108] developed specific spectral disease indices (SDIs) for the detection of sugar beet diseases. They tried to test all possible combinations of light wavelength to get the best-weighted variety on the hyperspectral imaging data set. The result of this study shows that the spectral disease indices can improve disease detection and crop monitoring in agriculture practice [108]. Asaari et al. [109] did the detection of plant responses to drought by hyperspectral images for drought-tolerant maize selection. Their study used the standard normal variate (SNV) method to reduce illumination effects due to hyperspectral images being easily affected by illumination. The different plant spectra were compared to assume the plant trait's variation. The result of this study successfully detects the drought stress responses of maize and plant recovery effects after rewatering [109]. Yang et al. [110] developed a convolutional neural network (CNN) model to analyze the hyperspectral images for estimating the cold damage of corn seedlings. This study shows that the spectral analysis result based on CNN modeling correlates with the chemical method result. The study provides a reference for plant cold damage study [110].

Although the hyperspectral image sensor is very promising for plant phenotyping applications, the limitations of hyperspectral image sensors still need to be considered. Currently, significant problems are the challenge of image processing, image analysis, and the heavy price of sensors [72], [80]. Many advanced methods like unsupervised learning [107], image fusion, artificial neural networks [105], [106] and



FIGURE 14. Depth image sensor principle.

convolutional neural network [110] may become potential solutions for the image processing and analysis problems.

E. DEPTH IMAGE SENSOR

A depth image sensor can provide images with real-time depth information. The current principle of depth camera can be divided into three parts: structure light, time-of-flight (ToF), and stereo depth. Fig. 14 presents the three depth image sensor principles. In the structure light method, the structural projector will emit patterned light on the object. Since the projected light pattern is known, the image sensor could provide depth information based on the deformation of pattern [111]. In the time-of-flight (ToF) method, the projector will emit light to the object. Because the speed of light is known, the time that light flight to an object and is reflected the image sensor will provide depth information [112]. Two image sensors (spaced a small distance apart) will be applied to take two images in the stereo depth method. Because the distance between the two image sensors is known, the comparison of two images will provide depth information [113]. Based on these principles, depth image sensors can produce an image showing the distance between the object and the image sensor [22]. The popular depth image sensor is currently an RGB-D image sensor equipped with a digital image, depth, and infrared emitters. Since RGB-D image sensor can simultaneously provide RGB and depth images of an object, it has been widely integrated into various applications such as 3D reconstruction, object recognition, and remote control [67]. Fig. 13 presents the image of depth image sensor.

IEEE INSTRUMENTATION & MEASUREMENT INTERPORT SOCIETY®



FIGURE 15. LIDAR [120], [121].

In the high-throughput plant phenotyping field, the RGB-D image sensor has shown its impressive ability for 3D plant phenotyping and automation of agricultural applications [115]. Seiffert et al. [67] applied Kinect RGB-D image sensor to capture RGB and depth images of plants for the segmentation of plant leaves. This study shows the success of 3D segmentation of individual plant leaves, which could be used for plant stress and disease resistance. Paulus et al. [115] proved the reliability of a low-cost RGB-D image sensor for plant traits (height, width, volume, surface, and compactness) measurement from the 3-D reconstruction of the plant. Bahman et al. [116] applied Intel RealSense D435 depth image sensor for non-invasive and automated plant height measurement. In this study, the depth image sensor was set up on the top of the plants. Image segmentation was applied to reduce the background impacts in plant height measurement. The plant height are calculated by using the distance of the depth camera from the ground subtracting the shortest length of the depth camera from the plant and the height of the pot. The result of this study shows high correlations and accuracy between depth image sensor measured plant height and actual plant height [116].

The limitations of depth image sensors are low resolution, short sensing distance, and sensitivity to optical interference [117].

F. LIDAR

Light detection and ranging (LiDAR) (also called laser scanner) is a popular remote sensing technique in many fields such as robotics, smartphones, transport, agriculture, etc. LiDAR can measure the distance by illuminating the object with laser light and measuring the reflection [118]. Differences in laser return time can be used to make digital 3-D representations of the thing. After computer processing, the 3-D data from LiDAR can provide an accurate estimated model of object [57]. In high-throughput plant phenotyping research, LiDAR is usually used for 3-D reconstruction of plants and estimation of plant's traits [64], [119]. Fig. 15 presents the image of LiDAR and pulsed time of flight reference design. Fig. 16 presents the LiDAR principle.

Eitel *et al.* [122] applied LiDAR for biomass and crop nitrogen estimation on wheat. The plant biomass and nitrogen concentration show a strong relationship with LiDAR-derived results; this proves the ability of LiDAR to improve crop nitrogen management in the wheat-growing section [122]. Jimenez-Berni *et al.* [123] mounted a LiDAR on a ground-based platform to estimate canopy height, ground cover, and above-ground biomass of maize. In their study, red reflectance image and canopy height were applied for the ground cover estimation, while the 3D voxel index (3DVI) and 3D profile index were used to estimate biomass. This study shows that all the estimated parameters have a strong relationship with LiDAR. Their work provides a more efficient and reliable method for plant phenotyping measurement [123].

Sun *et al.* [124] did a cotton plant growth analysis by using LiDAR. The LiDAR was implemented to measure plant height, canopy area, and plant volume. The result represents a strong relationship between manual measurements and LiDAR measurements [124]. Besides, [125] utilized the LiDAR data for corn plant. The leaf area, leaf distribution, and 3D model were automatically extracted from LiDAR point clouds by Difference of Normal (DoN). The result of this study shows overall 94.10% of the accuracy assessment [125]. In addition, [126] used LiDAR and other sensors to do a tree induction classification for the identification of post-harvest growth [126]. Sirault *et al.* [127] applied LiDAR to capture the structural dynamic of plant growth [127].

Although LiDAR provides a method to measure some plant traits accurately, some limitations need to be addressed. The end of LiDAR is that it has a high sensitivity to slight variations in path length [80]. Besides, high accurate LiDAR is very expensive [22]. According to [128], multiple LiDAR sensors might be a potential solution for the sensitivity problem because of their features of long-range and high resolution. Besides, the MEMS LiDAR sensors could be a good choice. About the high-cost problem, with the development of LiDAR manufacturing, high accurate LiDAR will gradually become affordable [128].

G. FLUORESCENCE SENSOR

Fluorescence sensors can measure chlorophyll by detecting fluorescence. Generally, a fluorescence sensor uses a charge-coupled device (CCD) sensitive to fluorescence signals. Since the plant chlorophyll will use the external light to excite the photosynthesis, the FLUO images can be captured by giving an external light excitation such as visible or UV (ultraviolet) light. The plant's metabolism will change when photosynthesis is affected by environmental conditions.







FIGURE 18. Fluorescence sensor [130].

Therefore, the fluorescence sensor can be used for plant stress and plant disease detection [57], [129]. Fig. 17 presents the fluorescence sensor principle. The control system sends the instruction to an external light emitter, which will emit visible or ultraviolet light to the surrounding environment. The plant will use external light to excite its photosynthesis. At the same time, the fluorescence sensor will measure chlorophyll data and send data to the process system to acquire the plant condition. Fig. 18 shows the image of the fluorescence sensor.

In [66] work, they applied a fluorescence sensor on the LemnaTec 3D Scanalyzer system. A fluorescence sensor captured the FLUO images through constant blue light excitation. These images could be applied for the detection of plant diseases like senescence, necrosis, and chlorosis [66]. Chen *et al.* [24] also used fluorescence sensor on the LemnaTec system. The FLUO images were applied to detect signals of chlorophyll fluorescence. The chlorophyll fluorescence excited by constant blue light can be used to describe the plant's health condition in a drought environment [24].

A fluorescence sensor is a powerful tool for plant phenotyping because it can measure the chlorophyll related to the plant's metabolism. However, the limitations of the fluorescence sensor are the requirement for intensive illumination and the view limitation [22]. Besides, most FLUO images are limited to single leaves or the seedling



FIGURE 19. AS7263 spectral sensor principle [132].

level of model crops. The research [57] suggested that a more robust data analysis software might be a potential solution for large-scale plant phenotyping. In addition, a standard procedure needs to be developed for fluorescence image processing to achieve large-scale plant phenotyping task [57].

H. LOW COST PLANT PHENOTYPING SENSORS1) AS7263 SPECTRAL SENSOR

The "ams multi-spectral sensor" product includes low-cost spectral sensors that can detect multiple channels in the visible and near-infrared spectrum. In these spectral sensors, the AS7263 near-infrared (NIR) spectral sensor is a low-cost plant phenotyping sensor based on the near-infrared spectrum principle. The AS7263 spectral sensor has six near the infrared channel, which can detect 610, 680, 730, 760, 810, and 860 nm wavelengths. Fig. 19 shows the AS7263 spectral sensor principle. Since the NIR lights could be greatly reflected by plant tissue, the AS7263 spectral sensor can be applied to detect the quality. Sripaurya et al. [131] implemented an AS7263 spectral sensor to do the banana quality measurement by attaching the sensor to the banana peel. The AS7263 spectral sensor will emit light to banana tissue by its LED driver. Then, the reflected light from banana tissue will pass through the sensor's six near the infrared channel. The reflected light intensity results are related to banana properties and maturity stages [131]. The study shows that low-cost multiple channels near-infrared (NIR) spectral sensor can be applied to estimate the quality parameters of banana fruits.

2) LOW-COST CHLOROPHYLL METER

Chlorophyll is important for the determination of health condition [133]. However, the current commercial chlorophyll meters (SPAD-502 chlorophyll meter, CL-01 chlorophyll meter, and MC-100 chlorophyll meter) are relatively expensive. There is a demand for a low-cost chlorophyll meter. Evan Hutomo *et al.* [134] developed a low-cost chlorophyll meter (LCCM) based on the light reflectance mechanism. The chlorophyll meter includes one near-infrared LED, and



FIGURE 20. Low-cost chlorophyll meter (LCCM) principle [134].

one red LED as resources of light. One TSL250 photodiode was used to convert light intensity to the output voltage. Fig. 20 shows the low-cost chlorophyll meter (LCCM) principle. The result of this study shows that the low-cost chlorophyll meter can measure the chlorophyll with a high coefficient of determination in correlation with the SPAD-502 chlorophyll meter's result [134].

3) FLIR LEPTON

The FLIR Lepton is a low-cost long-wave infrared image sensor for thermal image generation. The current FLIR Lepton has two versions (Lepton $3.5: 160 \times 120$ pixels and Lepton 2.5 : 80×60 pixels) and can easily integrate into multiple platforms [135]. Acorsi et al. [136] assessed the performance of the FLIR Lepton 3.5 in both ground and aerial situations based on the result from the commercial infrared thermometer. Their study shows that FLIR Lepton has comparable measurement performance to more expensive in-ground and aerial models. Besides, the study demonstrates that FLIR Lepton accuracy decreased towards increasing flight altitudes. Additional calibration can be the solution for measurement accuracy problems at high flight altitudes [136]. Bruno et al. [137] applied FLIR Lepton 3.5 integrated with a Raspberry Pi to generate thermal images of crops. The evaluation of yield thermal images can predict crop water stress. The study shows that FLIR Lepton 3.5 can work as a low-cost and non-invasive infrared image sensor for determining the status of crops.

4) RASPBERRY PI SERIES CAMERAS AND SENSORS

The Raspberry Pi series cameras are a series of low-cost cameras, which includes one Raspberry Pi Camera Module 2 and one Raspberry Pi Camera Module 2 with no infrared filter (NoIR). The module 2 camera has a Sony IMX219 8-megapixel image sensor and can be used to take RGB images. The module 2 NoIR camera includes the same Sony IMX219 8-megapixel image sensor but excludes an infrared filter. Therefore, the module 2 NoIR camera is suitable for capturing infrared images. Sangjan *et al.* [138] applied Raspberry Pi Module 2 and Module 2 NoIR cameras-based sensor system for automated in-field spring wheat plant phenotyping measurement. The study used low-cost sensor systems with dual cameras to capture RGB images and NoIR images and extract information. The image analysis calculated the plant height, vegetation index, and NDVI data.

The proposed sensors system provides a low-cost method for high-throughput plant phenotyping [138].

Table 2 presents the summary of the plant phenotyping sensors. Table 3 presents sensors related parameters.

IV. SENSING PLATFORMS FOR HIGH-THROUGHPUT PLANT PHENOTYPING

With the development of sensor technology, automatic control technology, aeronautics, and computer engineering, plant phenotyping platforms have significantly been applied to do plant phenotyping research. A plant phenotyping platform can facilitate high-throughput plant phenotyping study by providing an automatic measurement solution for multiple plant phenotyping parameters. Many research institutions have introduced or developed different plant phenotyping platforms to integrate various sensors. The current plant phenotyping platforms can be divided into two parts: groundbased phenotyping platform and aerial-based phenotyping platform [22], [57]. The following subsections will discuss different types of plant phenotyping platforms.

A. GROUND BASED PLANT PHENOTYPING PLATFORMS Ground-based phenotyping platforms include conveyors, vehicles, or robotics with sensing sensors. Various sensors are installed on different positions and heights for measuring multiple plant phenotyping. Besides, the power system, microcontroller, GPS receiver, encoder, and other accessories will be installed on the platform to help the whole system work successfully [144]. The ground-based plant phenotyping platforms can reduce human labor and improve work efficiency. It plays an essential role in high-throughput plant phenotyping application [22].

The ground-based plant phenotyping platforms are suitable for sensor deployment. Various sensors can be applied on platforms and achieve multiple plant phenotyping measurements. Besides, ground-based plant phenotyping platforms are ideal for GPS/GIS tagging, which can help the platform's position check and automatic work. Moreover, ground-based phenotyping platforms can capture plant data at the plot level, which makes the data in high spatial resolution with relatively simple post-processing [57].

The ground-based plant phenotyping platforms also have some limitations. It generally takes a long time to cover a field. Therefore, large-scale plant phenotyping measurements of plants in a short time are impossible with ground-based phenotyping platforms [57].

Companies or universities have developed many groundbased plant phenotyping platforms. The ground-based plant phenotyping platforms are summarized in Table 4.

1) LEMNATEC 3D SCANALYZER

LemnaTec designs LemnaTec 3D Scanalyzer platform for the quantitative, non-invasive analysis of large scale plants. Each plant can be continuously imaged by various image sensors, employing different spectra to get additional plant phenotyping information. The large-scale significant plant data will

Sensor Type	Operation Principle	Advantages	Applications	Limitations	Potential Solutions
Visible spectrum im- age sensor	Detect visible spec- trum (400 to 780 nm) to form the RGB im- age [22]	Low cost [70], Easy to handle [72], Ubiquitous [72]	Morphometric pa- rameters [74], Plant growth [74], Biomass [78]	Limited to visual spectral bands [80], Easily affected by the light condition [22]	Apply multiple im- age sensors plat- form, Digital image sen- sor scene mode [81]
Infrared spectrum im- age sensor	Detect infrared spec- trum (780 nm - 1000 µm) information [87]	Easy to handle [87] Reflectance is related to water content (NIR) [72] Accurate plant temperature measurement (TIR) [87]	Leaf water content (NIR) [24], Drought response (NIR) [86], Plant temperature (TIR) [90], Plant water stress (TIR) [91]	Limited to specific bands [72], Easily affected by environment condi- tion [80], Hard to detect the difference of small temperature (TIR) [80]	Apply multiple im- age sensors plat- form, Apply cooled de- vice (TIR) [87]
Multi-spectral image sensor	Detect the light with few discrete wide spectral bands [80]	Provide multi- spectral information [80]	Plant stress [96, 99], Disease resistance [97], NDVI [95]	Limited to few spec- tral bands [80]	Hyperspectral image sensor [80]
Hyperspectral image sensor	Detect the light with hundreds of narrow spectral bands [101]	Provides hyperspectral information [101]	Plant stress [107] [110], Plant assess- ment [105], Condi- tion monitor [108, 109]	Image processing [72], Image analysis [72], High cost [80]	Image fusion [105], Machine learning [106, 110]
Depth image sensor	Provides image with depth information [22]	Relative low cost [115], Provides depth information [115]	3-D reconstruction of plant to estimate plant traits [115], Plant stress [67]	Low resolution [117], Short sensing distance [117], Sensitive to optical interference [117]	Apply multi-depth image sensors [117], Apply image segmentation [116]
LiDAR	Generates digital 3-D representations of the plant [64]	High resolution [57], Highly accurate [57]	3-D reconstruction of plant to esti- mate plant traits [64, 119]	Hard to sense small variations [80], High cost [22]	Improve algorithm, Working with other sensors [126]
Fluorescence sensor	Measures the chloro- phyll by fluorescence [129]	Acquire plant metabolic information [129]	Plant stress [24], Disease resistance [66]	Require intensive illumination [22], View limitation [22], Does not work for large scale plants [57]	Powerful data anal- ysis software [57], Formulate standard procedure [57]

TABLE 2. Summary of the plant phenotyping sensors.

play an essential role in plant analysis. All plant data is available in the database within minutes of imaging for plant development. These comprehensive data will be helpful for physiological and genetic plant modeling, which can provide deep insights into plant phenomics and plant breeding [24], [66], [74], [86].

2) HIGH-THROUGHPUT RICE PHENOTYPING FACILITY

High-throughput rice phenotyping facility works on the rice traits. Group G1 will be transported to the conveyor from the greenhouse by an automatic guided vehicle on this platform. Then the G1 plants will be transported to the inspection unit, where the G1 plants are constantly measured by X-ray devices and image sensors. After that, another group, G2, will be swapped with G1 and measured by the inspection unit. The inspection unit applies a control software LabVIEW 8.6 (National Instruments, USA) for image capturing, image processing, and data communication [78].

3) PHENOVISION

PHENOVISION is a greenhouse system for high-throughput plant phenotyping of crops. Plant Pots are transported

through a conveyor belt system. The imaging cabins can apply up to three image sensor systems including visual spectrum image sensors, thermal infrared image sensor and stateof-the-art hyperspectral imaging system in a light-controlled conditions to get the plant phenotyping data [145].

4) LEEAGRA 3434 DL

The LeeAgra 3434 DL is a kind of ground vehicle. It has a height of 1.93 m and provides minimal impact to the ground plants. A boom is attached in the front of the tractor, where various sensors and instrument components are installed for plant phenotyping measurement [95].

5) LEICA SCANSTATION 2

Leica ScanStation 2 is a laser instrument. The LiDAR scanner (TLS) uses a 532 nm pulsed green laser. In the range of 0–50 m, the LiDAR scanner can work with a 4 mm beam diameter, sample density of 1 mm, and 50 kHz rate [122].

6) PHENOMOBILE LITE

Phenomobile Lite is an advanced phenotyping buggy with the advantages of being adaptable and easy to handle for



TABLE 3. Sensors attributes.

Sensor Type	Plant Parameters	Resolution/Range	Sensitivity	Cost (USD)
Visible spectrum im- age sensor	Plant height Leaf length/width/area/angle Canopy cover Biomass Plant stress	2056 \times 2454 pixels [74] 4056 \times 3040 pixels [139]	60 mm per pixel [74]	≈ 1000
Infrared spectrum im- age sensor	Plant/Water stress Leaf water content	14-bit radiometric resolu- tion (TIR) [91], 1280 × 1024 pixels (NIR) [140]	0.5 °C (TIR) [91], 35 mm per pixel (NIR) [140]	$\approx 375 \text{ (TIR)} \\ \approx 500 \text{ (NIR)}$
Multi-spectral image sensor	Plant/Water stress Leaf water content Disease detection NDVI Biomass	2048 × 1536 pixels [96] 1260 × 960 pixels [98]	60 mm per pixel [96] 35 mm per pixel [98]	≈ 2000
Hyperspectral image sensor	Plant height Plant/Water stress Leaf water content Disease detection Chlorophyll NDVI Biomass	510 × 328 pixels [109] 640 × 640 pixels [107] 510 × 328 pixels [109]	Spectral resolution of 4 nm [107] Spectral resolution of 2.8 nm [108] Spectral resolution of 3.1 nm [109]	≈ 2000
Depth image sensor	Plant height Leaf length/width/area/angle Canopy cover Biomass Plant stress	1280 × 720 pixels [141] 1280 × 800 pixels [142]	Error <2% at 2 m [141] Error <2% at 4 m [114]	≈ 200
LiDAR	Plant height Leaf length/width/area/angle Canopy cover Plant stress Biomass	100 m [143] 50 m [122] 80 m [124]	3 cm [143] 4 – 6 mm [122]	≈ 1800
Fluorescence sensor	Chlorophyll	0.01mg/L [130]	±0.2% FS [130]	$ \approx 500$

field plant phenotyping applications. It has a flexible space for different sensor equipment. This ground platform can provide a non-destructive high-throughput plant phenotyping solution by rapidly scanning field crops [146].

7) SPIDER DL

The Spider DL is a tractor platform for plant phenotyping work. It has large tires, which can provide minimal plant disturbance. Its lightweight (3300 lb) can reduce the damage to the field and make transportation easier [124].

8) TEXAS A&M GROUND PHENOTYPING VEHICLE

The Texas A&M Ground Phenotyping Vehicle is a ground vehicle for plant phenotyping. It can carry various sensors for multiple plant phenotyping data. The three meters height and high stability can essentially reduce the disturbance to the crops when working for corn, sorghum, and other row crops plant phenotyping measurement [65].

B. AERIAL BASED PLANT PHENOTYPING PLATFORMS

Aerial-based plant phenotyping platforms gradually increase since they can overcome some ground-based plant phenotyping platforms' limitations. It has been considered an essential component of plant phenotyping [147]. Aerial-based plant phenotyping platforms can enable the measurement of plant parameters from several plots within minutes. Typical aerial platforms can be divided into manned aerial vehicles

VOLUME 1, 2022

(MAVs), unmanned aerial vehicles (UAVs), and satellites. Currently, the more popular platform is UAVs since the satellites have the limitation of image resolution, and the MAVs need more manual operations. Unmanned aerial vehicles (UAVs), also known as a drone, is an aircraft that can do flight work without any humans on board. The flight of UAVs can be operated by a human operator or autopilot system. UAVs are excellent aerial-based platforms for taking high-quality and vast amounts of sensing data. Sensors such as visual spectrum image sensors, infrared image sensors, hyperspectral image sensors, multi-spectral image sensors, depth image sensors, and LiDAR systems can be mounted on UAVs for multiple plant phenotyping parameters measurements [80], [148]. Besides, with the rapid development of control technology, various drones can be applied by farmers who were trained in a short time.

The aerial-based platforms have a larger measuring area and better efficiency than ground-based platforms. Aerialbased plant phenotyping platforms can measure large-scale plants quickly, which is impossible on ground-based platforms. Therefore, the plant data collected by aerial-based media can represent large-scale plant conditions without any time delay [57].

However, aerial-based plant phenotyping platforms have a limitation of load. The payload of a UAV varies from 0.8 to 10 kg, which means it can not bring too many sensors on the platforms for multiple plant phenotyping measurements [80].

Diatform Nome	Imaga Sancara	Massured Dhanatuning Dayamatans	Defenences
Platform Name	image Sensors	Measured Phenotyping Parameters	References
LemnaTec 3D Scanalyzer	Visible spectrum image sensor, NIR	Plant height, Plant space, Biomass, Plant	[24, 66, 74, 86]
	image sensor	stress	
High-throughput rice phenotyping facil-	Visible spectrum image sensor	Plant height, Leaf area, Plant compact-	[78]
ity		ness, Biomass	
PHENOVISION	VNIR-HS image sensor	Plant stress	[109]
LeeAgra 3434 DL	Sonar proximity sensor, Apogee SI- 121 infrared radiometer (IRT) sensors, Crop Circle ACS-470 multi-spectral crop canopy sensor	Plant height,Temperature, Canopy reflectance-NDVI	[95]
Leica ScanStation 2	Time-of-flight, terrestrial LiDAR	Biomass, Crop nutritional status	[122]
Phenomobile Lite	LiDAR (SICK LMS 400-2000, SICK AG, Waldkirch, Germany), NDVI sen- sor (GreenSeeker, Trimble, USA), Visi- ble spectrum image sensor (Canon 6D, Canon Inc., Tokio, Japan)	Plant height, Canopy cover, Biomass	[123]
Spider DL	LiDAR (LMS 511 PRO SR, SICK AG, Waldkrich, Germany)	Plant height, Projected canopy area (PCA), Plant volume (PV)	[124]
Texas A&M Ground Phenotyping Vehi- cle	Senix 15S-485 Ultrasonic sensors, Crop Circle ACS-430 spectral sensors, Apogee SI-131 IRT sensors, Micro Ep- silon CTF-CF15 IRT sensors	Canopy height, NDVI, Canopy temper- ature	[65]

TABLE 4. Ground based plant phenotyping platforms.

TABLE 5. Aerial based plant phenotyping platforms.

Platform Name	Image Sensors	Measured Phenotyping Parameters	References
X8 Octocopter	Two Canon S110 point-and-shoot, Vi- sual spectrum image sensors	Canopy area, Plot row length	[77]
Tarot Iron Man 1000 octocopter	DIY-Thermocam	Canopy temperature, Water status	[91]
Airelectronics fixed-wing	Tetracam image sensor	NDVI, Plant stress	[96]
DJI Phantom 3 quadcopter	MicaSense RedEdge multi-spectral im- age sensor	Plant disease detection	[97]
DJI Inspire-1 Pro UAV	Digital Zenmuse X5 image sensor, Mi- caSense RedEdge image sensor	Plant/Water stress	[98]
Draganfly X4P quad-copter	MicaSense RedEdge image sensor	Plant stress	[99]
DJI M600	Sony alpha 7R, Velodyne VLP-16 Puck HI-RES LiDAR	Plant height, Canopy cover	[143]
UAVs	Headwall Photonics Nano-Hyperspec pushbroom scanner, Velodyne VLP-16 3D LiDAR sensor	Plant height, Canopy cover, Biomass	[152]

Besides, speed and altitude will influence the resolution of aerial-based platform sensors, which may lead to image loss [22].

Companies or universities have developed many aerialbased plant phenotyping platforms. The aerial-based plant phenotyping platforms are summarized in Table 5.

1) X8 OCTOCOPTER

X8 is an autonomously flying octocopter, which can carry a 1 kg payload with 15 minutes flight duration. The flight system uses GPS and onboard data to allow two control modes: full autopilot mode and manual mode [77].

2) TAROT IRON MAN 1000 OCTOCOPTER

The Tarot Iron Man 1000 octocopter is composed of carbon fibers. It has brushless 600 W power T-Motors for holding fiber propellers. Its payload capacity is 4 kg and the flight time is about 20 minutes [91].

3) AIRELECTRONICS FIXED-WING

Airelectronics fixed-wing is a UAV-based remote aerial sensor platform. It is a fixed-wing platform that can be controlled through an autopilot system. It also has the ability of waypoint navigation and altitude control. Its payload capacity is 1.5 kg and flight time is about 30 minutes [96].

4) DJI PHANTOM 3 QUADCOPTER

DJI Phantom 3 quadcopter system has a microcontroller, 10 DOF sensors, 3D printed supporting stand, and vibration dampeners. This system can calculate the angle of the image sensor through built-in accelerometers and gyroscopes. Besides, the altitude can be determined by the built-in barometer. When the mounted image sensor is stable and at the target height, the plant images will be captured [97].

5) DJI INSPIRE-1 PRO UAV

DJI Inspire-1 Pro UAV has a payload capacity of 3.5 kg, and its flight duration measured approximately 15 minutes.



It has various intelligent flight modes, including waypoint, home lock, point of interest, and course lock. These modes will be beneficial for getting better plant phenotyping data [98], [149].

6) DRAGANFLY X4P QUAD-COPTER

Draganfly X4P quadcopter is designed for line-of-sight flying. It has the ability of automatic take-off and intelligent flight navigation. Users can use normal flight mode or build-in software to control the quadcopter [99], [150].

7) DJI M600

The DJI Matrice 600 (M600) is an aerial platform designed for aerial photography and multiple industrial applications. It integrates various powerful DJI technologies, including intelligent flight control and battery management systems. Its payload capacity is 5.5 kg and flight time is about 15 minutes [143], [151].

V. SUMMARY AND FUTURE PERSPECTIVES

As discussed in this review paper, crop monitoring could be a potential solution to the problems of food shortage for growing world's population since it can provide up-to-date information regarding crops for farmers in support of increasing food production. High-throughput plant phenotyping is believed to be a crucial part of crop monitoring on the data acquisition of the large-scale crop characteristics. The related advanced technologies and sensors can facilitate the development of high-throughput phenotyping for crop monitoring. It is believed that with the development of new technologies, crop monitoring with high-throughput phenotyping will gradually become the central part of agriculture. Below is the summary of the significant challenges of high-throughput plant phenotyping and future perspectives:

- High accuracy and low-cost sensor solution for plant phenotyping measurement.
- The challenge of crop image processing and image analysis. Integration of advanced technologies like machine learning and image fusion may become the potential solution.
- The relationship between plant phenotyping and genomics needs to be further studied to understand higher-level breeding requirements and application potential.
- High speed, wide range, low energy consumption, and low-cost communication protocol need to be developed for the high-quality crop monitoring system.
- Currently available plant phenotyping system can be broadly divided into two categories:
 - (a) Plant phenotyping system;

(b) Plant phenotyping system + IoT based plant growth environment monitoring system;

• The development of integrated plant phenotyping systems for plant growth environment monitoring, plant phenotyping measurement, data transmission, data management, and data visualization will be a future research trend.

REFERENCES

- N. Gondchawar and R. Kawitkar, "IoT based smart agriculture," *Int. J. Adv. Res. Comput. Commun. Eng.*, vol. 5, no. 6, pp. 838–842, 2016.
- [2] R. Siche, "What is the impact of COVID-19 disease on agriculture?" *Scientia Agropecuaria*, vol. 11, no. 1, pp. 3–6, 2020.
- [3] R. Rayhana, G. Xiao, and Z. Liu, "Internet of Things empowered smart greenhouse farming," *IEEE J. Radio Freq. Identif.*, vol. 4, no. 3, pp. 195–211, 2020.
- [4] A. Goujon, "Human population growth," in *Encyclopedia of Ecology*. Amsterdam, The Netherlands: Elsevier, 2019, pp. 344–351.
- [5] J. Wang, T. He, and Y. Lin, "Changes in ecological, agricultural, and urban land space in 1984–2012 in China: Land policies and regional social-economical drivers," *Habitat Int.*, vol. 71, pp. 1–13, Jan. 2018.
- [6] Aqeel-ur-Rehman, A. Z. Abbasi, N. Islam, and Z. A. Shaikh, "A review of wireless sensors and networks' applications in agriculture," *Comput. Stand. Interfaces*, vol. 36, no. 2, pp. 263–270, 2014.
- [7] W. F. Abobatta, "Precision agriculture: A new tool for development," in *Precision Agriculture Technologies for Food Security and Sustainability*. Hershey, PA, USA: IGI Global, 2021, pp. 23–45.
- [8] N. Suma, S. R. Samson, S. Saranya, G. Shanmugapriya, and R. Subhashri, "IoT based smart agriculture monitoring system," *Int. J. Recent Innov. Trends Comput. commun.*, vol. 5, no. 2, pp. 177–181, 2017.
- [9] R. Rayhana, G. Xiao, and Z. Liu, "RFID sensing technologies for smart agriculture," *IEEE Instrum. Meas. Mag.*, vol. 24, no. 3, pp. 50–60, May 2021.
- [10] R. Rayhana, G. Xiao, and Z. Liu, "Printed sensor technologies for monitoring applications in smart farming: A review," *IEEE Trans. Instrum. Meas.*, vol. 70, no. 70, pp. 1–19, Sep. 2021.
- [11] M. Ayaz, M. Ammad-Uddin, Z. Sharif, A. Mansour, and E.-H. M. Aggoune, "Internet-of-Things (IoT)-based smart agriculture: Toward making the fields talk," *IEEE Access*, vol. 7, pp. 129551–129583, 2019.
- [12] R. Rayhana, Y. Jiao, Z. Liu, A. Wu, and X. Kong, "Real-time embedded system for valve detection in water pipelines," *J. Real-Time Image Process.*, vol. 19, pp. 247–259, Apr. 2022.
- [13] R. Rayhana, Y. Jiao, Z. Bahrami, Z. Liu, A. Wu, and X. Kong, "Smart valve detection system for water distribution networks," in *Proc. 22nd IEEE Int. Conf. Ind. Technol. (ICIT)*, vol. 1, 2021, pp. 1176–1181.
- [14] R. Rayhana, Y. Jiao, Z. Liu, A. Wu, and X. Kong, "Water pipe valve detection by using deep neural networks," in *Proc. Smart Struct. NDE Ind. 4.0 Smart Cities Energy Syst.*, vol. 11382, 2020, Art. no. 1138205.
- [15] M. A. Gehan and E. A. Kellogg, "High-throughput phenotyping," *Amer. J. Botany*, vol. 104, no. 4, pp. 505–508, 2017.
 [16] J. L. Araus and J. E. Cairns, "Field high-throughput phenotyping:
- [16] J. L. Araus and J. E. Cairns, "Field high-throughput phenotyping: The new crop breeding frontier," *Trends Plant Sci.*, vol. 19, no. 1, pp. 52–61, 2014.
- [17] R. Rayhana, Y. Jiao, A. Zaji, and Z. Liu, "Automated vision systems for condition assessment of sewer and water pipelines," *IEEE Trans. Autom. Sci. Eng.*, vol. 18, no. 4, pp. 1861–1878, Oct. 2021.
- [18] R. Rayhana, Y. Jiao, Z. Bahrami, Z. T. Liu, A. Wu, and X. Kong, "Valve detection for autonomous water pipeline inspection platform," *IEEE/ASME Trans. Mechatronics*, vol. 27, no. 2, pp. 1070–1080, Apr. 2022.
- [19] R. Rayhana, Z. Bahrami, T. Wang, Z. Liu, A. Wu, and X. Kong, "Environment prediction to enhance the navigation system of water pipeline inspection platforms," in *Proc. IEEE 30th Int. Symp. Ind. Electron. (ISIE)*, 2021, pp. 1–6.
- [20] F. Fiorani and U. Schurr, "Future scenarios for plant phenotyping," *Annu. Rev. Plant Biol.*, vol. 64, pp. 267–291, Feb. 2013.
- [21] D. K. Großkinsky, J. Svensgaard, S. Christensen, and T. Roitsch, "Plant phenomics and the need for physiological phenotyping across scales to narrow the genotype-to-phenotype knowledge gap," *J. Exp. Botany*, vol. 66, no. 18, pp. 5429–5440, 2015.
- [22] R. Qiu et al., "Sensors for measuring plant phenotyping: A review," Int. J. Agr. Biol. Eng., vol. 11, no. 2, pp. 1–17, 2018.
- [23] C. Granier and D. Vile, "Phenotyping and beyond: Modelling the relationships between traits," *Current Opin. Plant Biol.*, vol. 18, pp. 96–102, Apr. 2014.
- [24] D. Chen *et al.*, "Dissecting the phenotypic components of crop plant growth and drought responses based on high-throughput image analysis," *Plant Cell*, vol. 26, no. 12, pp. 4636–4655, 2014.
- [25] C. S. Bekkering, J. Huang, and L. Tian, "Image-based, organ-level plant Phenotyping for wheat improvement," *Agronomy*, vol. 10, no. 9, p. 1287, 2020.

- [26] S. Thapa, F. Zhu, H. Walia, H. Yu, and Y. Ge, "A novel LiDAR-based instrument for high-throughput, 3D measurement of morphological traits in maize and sorghum," Sensors, vol. 18, no. 4, p. 1187, 2018.
- R. Klose, J. Penlington, and A. Ruckelshausen, "Usability study [27] of 3D time-of-flight cameras for automatic plant phenotyping, Bornimer Agrartechnische Berichte, vol. 69, nos. 93-105, p. 12, 2009
- J. C. Rose, S. Paulus, and H. Kuhlmann, "Accuracy analysis of a [28] multi-view stereo approach for phenotyping of tomato plants at the organ level," Sensors, vol. 15, no. 5, pp. 9651-9665, 2015.
- U. Rascher et al., "Non-invasive approaches for phenotyping of [29] enhanced performance traits in bean," Funct. Plant Biol., vol. 38, no. 12, pp. 968-983, 2011.
- A. L. Thompson et al., "Deploying a proximal sensing cart to iden-[30] tify drought-adaptive traits in upland cotton for high-throughput phenotyping," *Front. Plant Sci.*, vol. 9, p. 507, Apr. 2018. Y. Savriama, "A step-by-step guide for geometric morphometrics of
- [31] floral symmetry," Front. Plant Sci., vol. 9, p. 1433, Oct. 2018.
- [32] M. E. Ghanem, H. Marrou, and T. R. Sinclair, "Physiological phenotyping of plants for crop improvement," Trends Plant Sci., vol. 20, no. 3, pp. 139-144, 2015.
- [33] P. P. Ray, "Internet of Things for smart agriculture: Technologies, practices and future direction," J. Ambient Intell. Smart Environ., vol. 9, no. 4, pp. 395-420, 2017.
- [34] E. Khorov, A. Kiryanov, A. Lyakhov, and G. Bianchi, "A tutorial on IEEE 802.11ax high efficiency WLANs," IEEE Commun. Surveys Tuts., vol. 21, no. 1, pp. 197-216, 1st Quart., 2019.
- G. R. Hiertz, D. Denteneer, L. Stibor, Y. Zang, X. P. Costa, and [35] B. Walke, "The IEEE 802.11 universe," IEEE Commun. Mag., vol. 48, no. 1, pp. 62-70, Jan. 2010.
- Y. Zheng and C. Liu, "Agricultural IoT system based on image [36] processing and cloud platform technology," in Proc. Int. Conf. Mach. Learn. Intell. Commun., 2017, pp. 33-42.
- J. Zhou et al. "CropQuant: An Automated and Scalable [37] Field Phenotyping Platform for Crop Monitoring and Trait Measurements to Facilitate Breeding and Digital Agriculture." bioRxiv. Jan. 2017. [Online]. Available: https://www.biorxiv.org/ content/10.1101/161547v2.full (Accessed: Jun. 14, 2022).
- R. Luca, P. Ciotîrnae, and V. Greu, "Evaluation and improvement [38] of WiMAX real throughput at application level using experimental study analysis and resulted optimization rules," in Proc. Int. Conf. Commun. (COMM), 2016, pp. 195-200.
- A. D. Boursianis et al., "Internet of Things (IoT) and agricultural [39] unmanned aerial vehicles (UAVs) in smart farming: A comprehensive review," Internet Things, vol. 18, May 2020, Art. no. 100187.
- [40] M. Usha Rani, C. Suganya, S. Kamalesh, and A. Sumithra, "An integration of wireless sensor network through Wi-Max for agriculture monitoring," in Proc. Int. Conf. Comput. Commun. Inf., 2014, pp. 1-5.
- H. M. Jawad, R. Nordin, S. K. Gharghan, A. M. Jawad, and [41] M. Ismail, "Energy-efficient wireless sensor networks for precision agriculture: A review," Sensors, vol. 17, no. 8, p. 1781, 2017.
- S. Kuhlgert et al., "Multispeq beta: A tool for large-scale plant [42] phenotyping connected to the open PhotosynQ network," Roy. Soc. Open Sci., vol. 3, no. 10, 2016, Art. no. 160592.
- [43] F. Amato, S. Amendola, and G. Marrocco, "Upper-bound performances of RFID epidermal sensor networks at 5G frequencies," in Proc. IEEE 16th Int. Conf. Wearable Implantable Body Sens. Netw. (BSN), 2019, pp. 1-4.
- J. G. Andrews et al., "What will 5G be?" IEEE J. Sel. Areas [44] Commun., vol. 32, no. 6, pp. 1065-1082, Jun. 2014.
- [45] T. Brown, C. Zimmermann, W. Panneton, N. Noah, and J. Borvitz, High-Resolution, Time-Lapse Inaging for Ecosystem-Scale Phenotyping in the Field. Totowa, NJ, USA: Humana Press, 2012, pp. 71-96. [Online]. Available: https://doi.org/10.1007/978-1-61779-995-2 7
- R. S. Sinha, Y. Wei, and S.-H. Hwang, "A survey on LPWA tech-[46] nology: LoRa and NB-IoT," ICT Exp., vol. 3, no. 1, pp. 14-21, 2017
- R. K. Singh, M. Aernouts, M. De Meyer, M. Weyn, and R. Berkvens, [47] "Leveraging LoRaWAN technology for precision agriculture in greenhouses," *Sensors*, vol. 20, no. 7, p. 1827, 2020.
- C. He, M. Shen, L. Liu, C. Okinda, J. Yang, and H. Shi, "Design and [48] realization of a greenhouse temperature intelligent control system based on NB-IoT," J. South China Agr. Univ., vol. 39, no. 2, pp. 117-124, 2018.
- [49] Adelantado, X. Vilajosana, P. Tuset-Peiro, B. Martinez, E. J. Melia-Segui, and T. Watteyne, "Understanding the limits of

LoRaWAN," IEEE Commun. Mag., vol. 55, no. 9, pp. 34-40, Sep. 2017

- [50] K. Mekki, E. Baijc, F. Chaxel, and F. Meyer, "A comparative study of LPWAN technologies for large-scale IoT deployment," ICT Exp., vol. 5, no. 1, pp. 1-7, 2019.
- T. M. Connolly and C. E. Begg, Database Systems: A Practical [51] Approach to Design, Implementation and Management. Boston, MA, USA: Pearson Educ., 2010.
- [52] N. Jatana, S. Puri, M. Ahuja, I. Kathuria, and D. Gosain, "A survey and comparison of relational and non-relational database," Int. J.
- J. Li and E. Xu, "Development on smart agriculture by wireless sensor networks," in *Proc. 1st Int. Conf. Ind. Fcon. Ind.* [53] 2015, pp. 41-47.
- [54] F. Schmidt, B. Bruns, T. Bode, H. Scharr, and A. B. Cremers, "A distributed information system for managing phenotyping mass data," in Massendatenmanagement der Agrar-und Ernährungswirtschaft-Erhebung-Verarbeitung-Nutzung. Bonn, Germany: Gesellschaft Informatik, 2013.
- D. Reynolds, J. Ball, A. Bauer, R. Davey, S. Griffiths, and [55] J. Zhou, "CropSight: A scalable and open-source information management system for distributed plant phenotyping and IoT-based crop management," *Gigascience*, vol. 8, no. 3, 2019, Art. no. giz009. K. Sambrekar and V. S. Rajpurohit, "Fast and efficient multi-
- [56] view access control mechanism for cloud based agriculture storage management system," Int. J. Cloud Appl. Comput., vol. 9, no. 1, pp. 33-49, 2019.
- L. Li, Q. Zhang, and D. Huang, "A review of imaging techniques [57] for plant phenotyping," Sensors, vol. 14, no. 11, pp. 20078-20111, 2014.
- [58] M. P. Pound et al., "Deep machine learning provides state-of-the-art performance in image-based plant phenotyping," Gigascience, vol. 6, no. 10, 2017, Art. no. gix083.
- F. Al-Turjman, "The road towards plant phenotyping via WSNs: An [59] overview," Comput. Electron. Agr., vol. 161, pp. 4–13, Jun. 019. S. A. Tsaftaris and C. Noutsos, "Plant phenotyping with low cost
- [60] digital cameras and image analytics," in Information Technologies in Environmental Engineering. Heidelberg, Germany: Springer, 2009, pp. 238-251.
- V. Sagan et al., "UAV-based high resolution thermal imaging for [61] vegetation monitoring, and plant phenotyping using ICI 8640 P, FLIR Vue pro R 640, and thermomap cameras," Remote Sens., vol. 11, no. 3, p. 330, 2019.
- R. Xu, C. Li, and A. H. Paterson, "Multispectral imaging and [62] unmanned aerial systems for cotton plant phenotyping," PLoS One, vol. 14, no. 2, 2019, Art. no. e0205083. J. Behmann *et al.*, "Specim IQ: Evaluation of a new, miniaturized
- [63] handheld hyperspectral camera and its application for plant phenotyping and disease detection," Sensors, vol. 18, no. 2, p. 441, 2018
- [64] Y. Lin, "LiDAR: An important tool for next-generation phenotyping technology of high potential for plant phenomics?" Comput. Electron. Agr., vol. 119, pp. 61-73, Nov. 2015.
- S. C. Murray et al., "High clearance phenotyping systems for [65] season-long measurement of corn, sorghum and other row crops to complement unmanned aerial vehicle systems," in Proc. Auton. Air Ground Sens. Syst. Agr. Optim. Phenotyping, vol. 9866, 2016, Art. no. 986607.
- A. Hairmansis, B. Berger, M. Tester, and S. J. Roy, "Image-[66] based phenotyping for non-destructive screening of different salinity tolerance traits in rice," Rice, vol. 7, no. 1, pp. 1-10, 2014.
- C. Xia, L. Wang, B.-K. Chung, and J.-M. Lee, "In situ 3D segmenta-[67] tion of individual plant leaves using a RGB-D camera for agricultural automation," Sensors, vol. 15, no. 8, pp. 20463–20479, 2015. D. Sliney, "What is light? The visible spectrum and beyond," Eye,
- [68]
- Y. Zhu, X. Lei, K. X. Wang, and Z. Yu, "Compact CMOS spectral sensor for the visible spectrum," *Photon. Res.*, vol. 7, no. 9, [69] pp. 961-966, 2019.
- [70] A.-K. Mahlein, "Plant disease detection by imaging sensors-parallels and specific demands for precision agriculture and plant phenotyping," Plant Dis., vol. 100, no. 2, pp. 241-251, 2016.
- K. Mochida et al., "Computer vision-based phenotyping for improve-[71] ment of plant productivity: A machine learning perspective," *GigaScience*, vol. 8, no. 1, 2019, Art. no. giy153. N. Fahlgren, M. A. Gehan, and I. Baxter, "Lights, camera, action:
- [72] High-throughput plant phenotyping is ready for a close-up," Current Opin. Plant Biol., vol. 24, pp. 93-99, Apr. 2015. [Online]. Available: http://www.sciencedirect.com/science/article/pii/S1369526615000266



- [73] L. Hofer. "CMOS vs CCD Sensors and Overview." 2016. [Online]. Available: https://dataray.com/blogs/dataray-blog/cmos-vsccd-sensors-and-overview (Accessed: Jan. 20, 2021).
- N. Honsdorf, T. J. March, B. Berger, M. Tester, and K. Pillen, [74] "High-throughput phenotyping to detect drought tolerance QTL in wild barley introgression lines," *PLoS One*, vol. 9, no. 5, 2014, Art. no. e97047.
- E. H. Neilson, A. M. Edwards, C. Blomstedt, B. Berger, B. L. Møller, [75] and R. M. Gleadow, "Utilization of a high-throughput shoot imaging system to examine the dynamic phenotypic responses of a C4 cereal crop plant to nitrogen and water deficiency over time," J. Exp. Botany, Vol. 66, no. 7, pp. 1817–1832, 2015. M. R. Golzarian *et al.*, "Accurate inference of shoot biomass from
- [76] high-throughput images of cereal plants," Plant Methods, vol. 7, no. 1, p. 2, 2011.
- N. Yu, L. Li, N. Schmitz, L. F. Tian, J. A. Greenberg, and B. W. Diers, [77] "Development of methods to improve soybean yield estimation and predict plant maturity with an unmanned aerial vehicle based platform," Remote Sens. Environ., vol. 187, pp. 91-101, Dec. 2016.
- W. Yang *et al.*, "Combining high-throughput phenotyping and genome-wide association studies to reveal natural genetic variation [78] in rice," Nat. Commun., vol. 5, p. 5087, Oct. 2014.
- M. Klodt, K. Herzog, R. Töpfer, and D. Cremers, "Field phenotyping [79] of grapevine growth using dense stereo reconstruction," BMC Bioinf., vol. 16, no. 1, p. 143, 2015. S. Sankaran *et al.*, "Low-altitude, high-resolution aerial imaging
- [80] systems for row and field crop phenotyping: A review," Eur. J. Agron., vol. 70, pp. 112-123, Oct. 2015.
- I. Korobiichuk, V. Lysenko, O. Opryshko, D. Komarchyk, [81] N. Pasichnyk, and A. Juś, "Crop monitoring for nitrogen nutrition level by digital camera," in Proc. Conf. Autom., 2018, pp. 595-603.
- S.-R. Tsai and M. R. Hamblin, "Biological effects and medical applications of infrared radiation," J. Photochem. Photobiol. B Biol., [82] vol. 170, pp. 197-207, 2017.
- "Healthy Crop." Midopt. [Online]. Available: https://midopt.com/ [83] healthy-crop (Accessed: Jan. 20, 2021).
- "1.3 MP Custom Lens NIR Camera Module (monochrome)." E-Con [84] Systems. [Online]. Available: https://www.e-consystems.com/1MP-Customlens-NIR-Camera-module.asp (Accessed: Jan. 20, 2021).
- H-D. Seelig, A. Hoehn, L. Stodieck, D. Klaus, W. Adams, and W. Emery, "The assessment of leaf water content using leaf reflectance ratios in the visible, near-, and short-wave-infrared," *Int.* [85] J. Remote Sens., vol. 29, no. 13, pp. 3701-3713, 2008.
- [86] N. Briglia, G. Montanaro, A. Petrozza, S. Summerer, F. Cellini, and V. Nuzzo, "Drought phenotyping in Vitis vinifera using RGB and NIR imaging," Scientia Horticulturae, vol. 256, Oct. 2019, Art. no. 108555.
- R. Vadivambal and D. S. Jayas, "Applications of thermal imaging in [87] agriculture and food industry-A review," Food Bioprocess Technol., vol. 4, no. 2, pp. 186-199, 2011.
- J. Martínez, G. Egea, J. Agüera, and M. Pérez-Ruiz, "A cost-effective [88] canopy temperature measurement system for precision agriculture: A case study on sugar beet," Precis. Agr., vol. 18, no. 1, pp. 95-110, 2017.
- "DALI Infrared Thermal Imaging Temperature Sensor Factory Heat [89] Detector." [Online]. Available: https://www.alibaba.com (Accessed: Jan. 20, 2021).
- D. A. Matos, B. J. Cole, I. P. Whitney, K. J.-M. MacKinnon, [90] S. A. Kay, and S. P. Hazen, "Daily changes in temperature, not the circadian clock, regulate growth rate in Brachypodium distachyon,' PLoS One, vol. 9, no. 6, 2014, Art. no. e100072.
- L. G. T. Crusiol *et al.*, "UAV-based thermal imaging in the assessment of water status of soybean plants," *Int. J. Remote Sens.*, vol. 41, no. 9, [91] pp. 3243-3265, 2020.
- G. Kitić et al., "A new low-cost portable multispectral optical device [92] for precise plant status assessment," Comput. Electron. Agr., vol. 162, "Mutispectral Sensor." Idil. [Online]. Available: https://www.idil-
- [93] fibres-optiques.com/product/multispectral-sensor-pixelsensor (Accessed: Jan. 20, 2021).
- "Multispectral Camera for NDVI Crop Monitoring Sequoia+ Multispectral Camera." Parrot. [Online]. Available: https://www. [94] alibaba.com (Accessed: Jan. 20, 2021).
- P. Andrade-Sanchez et al., "Development and evaluation of a field-[95] based high-throughput phenotyping platform," *Funct. Plant Biol.*, vol. 41, no. 1, pp. 68–79, 2014.M. Zaman-Allah *et al.*, "Unmanned aerial platform-based multi-
- [96] spectral imaging for field phenotyping of maize," Plant Methods, vol. 11, no. 1, pp. 1-10, 2015.

- [97] A. Patrick, S. Pelham, A. Culbreath, C. C. Holbrook, I. J. De Godoy, and C. Li, "High throughput phenotyping of tomato spot wilt disease in peanuts using unmanned aerial systems and multispectral imaging," *IEEE Instrum. Meas. Mag.*, vol. 20, no. 3, pp. 4–12, Jun. 2017.
- [98] A. Kumar et al., "CIG based stress identification method for maize crop using UAV based remote sensing," in Proc. IEEE Sens. Appl. Symp. (SAS), 2020, pp. 1-6.
- S. Mardanisamani *et al.*, "Crop lodging prediction from UAV-[99] acquired images of wheat and Canola using a DCNN augmented with handcrafted texture features," in Proc. IEEE/CVF Conf. Comput. Vis. Pattern Recognit. Workshops (CVPRW), 2019, pp. 2657-2664.
- [100] P. Shippert, "Introduction to hyperspectral image analysis," J. Space *Commun.*, vol. 2, no. 3, p. 8, 2003. [101] A. Moghimi, C. Yang, and P. M. Marchetto, "Ensemble feature
- selection for plant phenotyping: A journey from hyperspectral to multispectral imaging," *IEEE Access*, vol. 6, pp. 56870–56884, 2018.
- [102] M. S. M. Asaari et al., "Close-range hyperspectral image analysis for the early detection of stress responses in individual plants in a high-throughput phenotyping platform," ISPRS J. Photogrammetry Remote Sens., vol. 138, pp. 121–138, Apr. 2018. [103] "Affordable, Powerful, Compact Hyperspectral Imager." LDP
- LLC. [Online]. Available: https://maxmax.com/maincamerapage/ hyperspectral (Accessed: Jan. 20, 2021).
- "Pika XC2r." RESONON. [Online]. Available: https://resonon.com/ [104] Pika-XC2 (Accessed: Jan. 20, 2021).
- [105] U. Seiffert, F. Bollenbeck, H.-P. Mock, and A. Matros, "Clustering of crop phenotypes by means of hyperspectral signatures using artificial neural networks," in Proc. 2nd Workshop Hyperspectral Image Signal Process. Evol. Remote Sens., 2010, pp. 1-4.
- [106] A. Backhaus, F. Bollenbeck, and U. Seiffert, "Robust classification of the nutrition state in crop plants by hyperspectral imaging and artificial neural networks," in Proc. 3rd Workshop Hyperspectral Image Signal Process. Evol. Remote Sens. (WHISPERS), 2011, pp. 1-4.
- [107] C. Römer et al., "Early drought stress detection in cereals: Simplex volume maximisation for hyperspectral image analysis," Funct. Plant *Biol.*, vol. 39, no. 11, pp. 878–890, 2012. [108] A.-K. Mahlein *et al.*, "Development of spectral indices for detecting
- and identifying plant diseases," Remote Sens. Environ., vol. 128, pp. 21-30, Jan. 2013.
- [109] M. S. M. Asaari, S. Mertens, S. Dhondt, N. Wuyts, and P. Scheunders, "Detection of plant responses to drought using close-range hyperspectral imaging in a high-throughput phenotyping platform,' in Proc. 9th Workshop Hyperspectral Image Signal Process. Evol. Remote Sens. (WHISPERS), 2018, pp. 1–5. W. Yang, C. Yang, Z. Hao, C. Xie, and M. Li, "Diagnosis of
- [110] plant cold damage based on hyperspectral imaging and convolutional neural network," IEEE Access, vol. 7, pp. 118239-118248, 2019
- [111] H.-H. Wu and A. Bainbridge-Smith, Advantages of Using a Kinect Camera in Various Applications, Univ. Canterbury, Christchurch, New Zealand, 2011, pp. 1-4.
- [112] T.-N. Nguyen, H.-H. Huynh, and J. Meunier, "3D reconstruction with time-of-flight depth camera and multiple mirrors," IEEE Access, vol. 6, pp. 38106-38114, 2018.
- [113] M. Carfagni et al., "Metrological and critical characterization of the Intel D415 stereo depth camera," Sensors, vol. 19, no. 3, p. 489, 2019
- [114] "Depth Camera D455." Intel. [Online]. Available: https://www. intelrealsense.com/depth-camera-d455/ (Accessed: Apr. 1, 2021).
- [115] S. Paulus, J. Behmann, A.-K. Mahlein, L. Plümer, and H. Kuhlmann, "Low-cost 3D systems: Suitable tools for plant phenotyping, Sensors, vol. 14, no. 2, pp. 3001–3018, 2014.
- [116] L. Bahman, "Height measurement of basil crops for smart irrigation applications in greenhouses using commercial sensors," M.S. thesis, Dept. Electr. Comput. Eng., Univ. Western Ontario, London, ON, Canada, 2019.
- [117] A. Kadambi, A. Bhandari, and R. Raskar, "3D depth cameras in vision: Benefits and limitations of the hardware," in Computer Vision and Machine Learning With RGB-D Sensors. Cham, Switzerland: Springer, 2014, pp. 3–26. R. Collis, "Lidar," Appl. Opt., vol. 9, no. 8, pp. 1782–1788, 1970.
- [118]
- [119] Q. Guo et al., "Crop 3D-A LiDAR based platform for 3D highthroughput crop phenotyping," Sci. China Life Sci., vol. 61, no. 3, pp. 328-339, 2018.
- [120] "LIDAR Pulsed Time of Flight Reference Design." Texas Instruments. [Online]. Available: https://www.ti.com/tool/TIDA-00663 (Accessed: Jan. 20, 2021).

- [121] "Hokuyo UST-10LX Scanning Laser Rangefinder." Robotshop. [Online]. Available: https://www.robotshop.com (Accessed: Jan. 20, 2021).
- [122] J. U. Eitel, T. S. Magney, L. A. Vierling, T. T. Brown, and D. R. Huggins, "LiDAR based biomass and crop nitrogen estimates for rapid, non-destructive assessment of wheat nitrogen status," *Field Crops Res.*, vol. 159, pp. 21–32, Mar. 2014.
- [123] J. A. Jimenez-Berni *et al.*, "High throughput determination of plant height, ground cover, and above-ground biomass in wheat with LiDAR," *Front. Plant Sci.*, vol. 9, p. 237, Feb. 2018.
- [124] S. Sun *et al.*, "In-field high throughput phenotyping and cotton plant growth analysis using LiDAR," *Front. Plant Sci.*, vol. 9, p. 16, Jan. 2018.
- [125] W. Su, M. Zhang, J. Liu, and Z. Sun, "Automated extraction of corn leaf points from unorganized terrestrial LiDAR point clouds," *Int. J. Agr. Biol. Eng.*, vol. 11, no. 3, pp. 166–170, 2018.
- [126] K. Koenig, B. Höfle, M. Hämmerle, T. Jarmer, B. Siegmann, and H. Lilienthal, "Comparative classification analysis of post-harvest growth detection from terrestrial LiDAR point clouds in precision agriculture," *ISPRS J. Photogrammetry Remote Sens.*, vol. 104, pp. 112–125, Jun. 2015.
- [127] X. Sirault *et al.*, "PlantScan: A three-dimensional phenotyping platform for capturing the structural dynamic of plant development and growth," in *Proc. 7th Int. Conf. Funct. Struct. Plant Models*, Saariselkä, Finland, 2013, pp. 45–48.
- [128] F. Zhao, H. Jiang, and Z. Liu, "Recent development of automotive LiDAR technology, industry and trends," in *Proc. 11th Int. Conf. Digit. Image Process. (ICDIP)*, vol. 11179, 2019, Art. no. 111794A.
- [129] C. Rousseau *et al.*, "High throughput quantitative phenotyping of plant resistance using chlorophyll fluorescence image analysis," *Plant Methods*, vol. 9, no. 1, p. 17, 2013.
- [130] "Online Fluorescence Do Sensor RS485 DO Probe Optical Dissolved Oxygen Sensor." REMOND. [Online]. Available: https://www.alibaba.com (Accessed: Jan. 22, 2021).
- [131] T. Sripaurya, K. Sengchuai, A. Booranawong, and K. Chetpattananondh, "Gros Michel banana soluble solids content evaluation and maturity classification using a developed portable 6 channel NIR device measurement," *Measurement*, vol. 173, Mar. 2021, Art. no. 108615.
- [132] "AS7263 6-Ch NIR Spectral Sensing Engine." ams. [Online]. Available: https://ams.com/as7263 (Accessed: May 1, 2022).
- [133] M. M. Shaahan, A. A. El-Sayed, and E. A. A Abou El-Nour, "Predicting nitrogen, magnesium and iron nutritional status in some perennial crops using a portable chlorophyll meter," *Scientia Horticulturae*, vol. 82, nos. 3–4, pp. 339–348, 1999.
- [134] E. P. Evan Hutomo, M. A. S. Adibawa, K. R. Prilianti, H. Heriyanto, and T. H. P. Brotosudarmo, "Low-cost chlorophyll meter (LCCM): Portable measuring device for leaf chlorophyll," in *Proc. 2nd Int. Seminar Photon. Opt. Appl. (ISPhOA)*, vol. 10150, 2016, Art. no. 101500V.
- [135] "LWIR Micro Thermal Camera Module-Lepton." Teledyne FLIR. [Online]. Available: https://www.flir.com/products/lepton/?model=3. 5+Lepton&vertical=lwir&segment=oem (Accessed: May 1, 2022).
- [136] M. G. Acorsi, L. M. Gimenez, and M. Martello, "Assessing the performance of a low-cost thermal camera in proximal and aerial conditions," *Remote Sens.*, vol. 12, no. 21, p. 3591, 2020.
- [137] H. Bruno, M. G. Narciso, G. O. Silva, and M. A. de Souza, "Lowcost thermal imager for plant phenotyping," in *Embrapa Arroz e Feijão-Resumo em anais de congresso (ALICE)*, 2017, p. 43.
- [138] W. Sangjan, A. H. Carter, M. O. Pumphrey, V. Jitkov, and S. Sankaran, "Development of a raspberry Pi-based sensor system for automated in-field monitoring to support crop breeding programs," *Inventions*, vol. 6, no. 2, p. 42, 2021.
- [139] "OpenCV OAK-1." OpenCV. [Online]. Available: https:// store.opencv.ai/products/oak-1 (Accessed: Apr. 1, 2021).
- [140] "IDS Imaging UI-1240ML-NIR Camera," 1stVision. [Online]. Available: https://www.1stvision.com/cameras/models/IDS-Imaging/ UI-1240ML-NIR (Accessed: Jan. 22, 2021).
- [141] "Depth Camera D435." Intel. [Online]. Available: https://www. intelrealsense.com/depth-camera-d435/ (Accessed: Apr. 1, 2021).
- [142] "OpenCV OAK-D." OpenCV. [Online]. Available: https://https:// store.opencv.ai/products/oak-d/ (Accessed: Apr. 1, 2021).
- [143] R. Ravi, Y.-J. Lin, T. Shamseldin, M. Elbahnasawy, M. Crawford, and A. Habib, "Implementation of UAV-based Lidar for high throughput phenotyping," in *Proc. IEEE Int. Geosci. Remote Sens. Symp.* (*IGARSS*), Jul. 2018, pp. 8761–8764.
- [144] J. Enciso, M. Maeda, J. Landivar, J. Jung, and A. Chang, "A ground based platform for high throughput phenotyping," *Comput. Electron. Agr.*, vol. 141, pp. 286–291, Sep. 2017.

- [145] "Phenovision Information." Plant Systems Biology. [Online]. Available: https://www.psb.ugent.be/phenotyping/phenovision/2phenovision (Accessed: Feb. 7, 2021).
- [146] "Phenomobile Lite Technology is a Mobile Ground-Based Platform for the Phenotyping of Crops." CSIRO. [Online]. Available: https://www.csiro.au/en/Do-business/Commercialisation/ Marketplace/phenoMobile (Accessed: Feb. 7, 2021).
- [147] F. Liebisch, N. Kirchgessner, D. Schneider, A. Walter, and A. Hund, "Remote, aerial phenotyping of maize traits with a mobile multisensor approach," *Plant Methods*, vol. 11, no. 1, p. 9, 2015.
- [148] X. Jin *et al.*, "High-throughput estimation of crop traits: A review of ground and aerial phenotyping platforms," *IEEE Geosci. Remote Sens. Mag.*, vol. 9, no. 1, pp. 200–231, Mar. 2021.
- [149] "INSPIRE 1 PRO." DJI. [Online]. Available: https://www.dji.com/ ca/inspire-1-pro-and-raw (Accessed: Feb. 7, 2021).
- [150] "Quadcopters and Multirotors." DRAGANFLYER. [Online]. Available: https://draganfly.com/products/quadcopters-multirotors (Accessed: Feb. 7, 2021).
- [151] "MATRICE 600." DJI. [Online]. Available: https://www.dji.com/ca/ (Accessed: Feb. 7, 2021).
- [152] A. Masjedi, N. R. Carpenter, M. M. Crawford, and M. R. Tuinstra, "Prediction of sorghum biomass using UAV time series data and recurrent neural networks," in *Proc. IEEE/CVF Conf. Comput. Vis. Pattern Recognit. Workshops (CVPRW)*, 2019, pp. 2695–2702.



ZHENYU MA received the B.Eng. degree from the Beijing University of Chemical Technology, China, and University of Detroit Mercy, USA. He is currently pursuing the M.A.Sc. degree in electrical engineering with The University of British Columbia, Kelowna, BC, Canada.



RAKIBA RAYHANA (Student Member, IEEE) received the B.Sc. degree in electrical and electronic engineering from BRAC University, Dhaka, Bangladesh, in 2015, and the M.Sc. degree in electronic and electrical engineering from the University of Strathclyde, Glasgow, U.K., in 2017. She is currently pursuing the Ph.D. degree in electrical engineering with The University of British Columbia, Kelowna, BC, Canada. Her research interest lies in precision agriculture, IoTs, sensors, embedded and intelligent systems, automation,

image processing, machine learning, and deep learning. She is also a student member of The International Society for Optics and Photonics Organization.



KE FENG received the Ph.D. degree in mechanical engineering from the University of New South Wales, Australia, in 2021. He is a Postdoctoral Research Fellow with The University of British Columbia, Kelowna, BC, Canada. He was a Postdoctoral Research Associate with the University of Technology Sydney, Australia, in 2021. His main research interests include digital-twin-based RUL prediction, vibration analysis, structural health monitoring, dynamics, tribology, signal processing, and machine learning.





ZHENG LIU (Senior Member, IEEE) received the Doctoral degree in engineering (earth resources) from Kyoto University, Kyoto, Japan, in 2000, and the Ph.D. degree in electrical engineering from the University of Ottawa, Canada, in 2007. From 2000 to 2001, he was a Research Fellow with the Nanyang Technological University, Singapore. He then joined the National Research Council of Canada, Ottawa, ON, Canada, as a Governmental Laboratory Visiting Fellow nominated by NSERC in 2001. Since 2002, he has been a Research

Officer associated two research institutes of NRC (Aerospace [IAR] & Construction [IRC]). From 2012 to 2015, he worked as a Full Professor with Toyota Technological Institute, Nagoya, Japan. He is currently with the Faculty of Applied Science, University of British Columbia-Okanagan as a Professor. His research interests include predictive maintenance, data/information fusion, computer/machine vision, machine learning, smart sensor and industrial IoT, and nondestructive inspection and evaluation. He is co-chairing the IEEE Instrumentation and Measurement Society technical committee (TC-1). He holds a Professional Engineer license in both British Columbia and Ontario. He serves on the editorial boards for journals including, IEEE TRANSACTIONS ON INSTRUMENTATION AND MEASUREMENT, IEEE/ASME TRANSACTIONS ON MECHATRONICS, IEEE JOURNAL OF RADIO FREQUENCY IDENTIFICATION, Information Fusion (Elsevier), Machine Vision and Applications (Springer), and IET/CAAI Transactions on Intelligence Technology. He also serves the Standards & Interoperability Committee of the Research Data Canada. He is a Fellow of SPIE.



YUEFENG RUAN received the Ph.D. degree in Germany. He is a Senior Research Scientist and a Durum Wheat Breeder with the Swift Current Research and Development Centre, Agriculture and Agri-Food Canada. He has registered eight durum wheat cultivars, including the first Intermediate Fusarium Head Blight resistance cultivar, and the first wheat midge and wheat stem sawfly insect resistance cultivar in Canada. He has published hundreds of miscellaneous publications in wheat breeding and research. His research and

breeding program develops new high quality and high yielding durum wheat cultivars with wide adaptability and improved profitability. As an active member in numerous Government and non-Government committees in Canada and worldwide, he provides the leadership in durum wheat research and development.



GAOZHI (GEORGE) XIAO (Fellow, IEEE) received the Ph.D. degree from Loughborough University, U.K., in 1995. He is a Principle Research Officer with the National Research Council Canada. He has managed large R&D projects in industries, academics and government labs covering areas including RFID/NFC, flexible/printable/wearable electronics, fiber optic sensor systems, photonic sensing and measurement, structural health monitoring, indoor air quality monitoring, structural materials, and smart materials. He has brought

several technologies from concepts to products. He has published more than 100 papers in refereed journals and refereed conference proceedings, and 1 co-edited book. He is currently holding 5 U.S. patents. He received the 2014 Technical Award from the IEEE Instrumentation and Measurement Society. He is currently the Editor-in-Chief of IEEE JOURNAL OF RADIO FREQUENCY IDENTIFICATION and an Associate Editor-in-Chief of IEEE TRANSACTIONS ON INSTRUMENTATION AND MEASUREMENT.



JATINDER S. SANGHA received the Ph.D. degree from the University of the Philippines, Los Baños/International Rice Research Institute, Philippines, in 2006. He is a Research Scientist with the Swift Current Research and Development Centre, Agriculture and Agri-Food Canada. His research statements are developing a top-ranked wheat physiology lab for fundamental and applied research for improving cereal quality and yield through efficient phenotyping and physiological understanding of traits contributing to improving

crop performance in diverse environments.